

### 

**Citation:** Croucher NJ, Mostowy R, Wymant C, Turner P, Bentley SD, Fraser C (2016) Horizontal DNA Transfer Mechanisms of Bacteria as Weapons of Intragenomic Conflict. PLoS Biol 14(3): e1002394. doi:10.1371/journal.pbio.1002394

Academic Editor: Nick H. Barton, Institute of Science and Technology Austria (IST Austria), AUSTRIA

Received: November 30, 2015

Accepted: January 29, 2016

Published: March 2, 2016

**Copyright:** © 2016 Croucher et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Sequence data have been deposited in Genbank with accession codes KT337339- KT337372.

**Funding:** NJC is funded by a Sir Henry Dale Fellowship, jointly funded by the Wellcome Trust and Royal Society (Grant Number 104169/Z/14/Z). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing Interests:** The authors have declared that no competing interests exist.

**RESEARCH ARTICLE** 

# Horizontal DNA Transfer Mechanisms of Bacteria as Weapons of Intragenomic Conflict

## Nicholas J. Croucher<sup>1</sup>\*, Rafal Mostowy<sup>1</sup>, Christopher Wymant<sup>1</sup>, Paul Turner<sup>2,3</sup>, Stephen D. Bentley<sup>4</sup>, Christophe Fraser<sup>1</sup>

 Department of Infectious Disease Epidemiology, Imperial College London, London, United Kingdom,
 Cambodia Oxford Medical Research Unit, Angkor Hospital for Children, Siem Reap, Cambodia, 3 Centre for Tropical Medicine and Global Health, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 4 Pathogen Genomics, Wellcome Trust Sanger Institute, Wellcome Trust Genome Campus, Hinxton, Cambridge, United Kingdom

\* n.croucher@imperial.ac.uk

### Abstract

Horizontal DNA transfer (HDT) is a pervasive mechanism of diversification in many microbial species, but its primary evolutionary role remains controversial. Much recent research has emphasised the adaptive benefit of acquiring novel DNA, but here we argue instead that intragenomic conflict provides a coherent framework for understanding the evolutionary origins of HDT. To test this hypothesis, we developed a mathematical model of a clonally descended bacterial population undergoing HDT through transmission of mobile genetic elements (MGEs) and genetic transformation. Including the known bias of transformation toward the acquisition of shorter alleles into the model suggested it could be an effective means of counteracting the spread of MGEs. Both constitutive and transient competence for transformation were found to provide an effective defence against parasitic MGEs; transient competence could also be effective at permitting the selective spread of MGEs conferring a benefit on their host bacterium. The coordination of transient competence with cellcell killing, observed in multiple species, was found to result in synergistic blocking of MGE transmission through releasing genomic DNA for homologous recombination while simultaneously reducing horizontal MGE spread by lowering the local cell density. To evaluate the feasibility of the functions suggested by the modelling analysis, we analysed genomic data from longitudinal sampling of individuals carrying Streptococcus pneumoniae. This revealed the frequent within-host coexistence of clonally descended cells that differed in their MGE infection status, a necessary condition for the proposed mechanism to operate. Additionally, we found multiple examples of MGEs inhibiting transformation through integrative disruption of genes encoding the competence machinery across many species, providing evidence of an ongoing "arms race." Reduced rates of transformation have also been observed in cells infected by MGEs that reduce the concentration of extracellular DNA through secretion of DNases. Simulations predicted that either mechanism of limiting transformation would benefit individual MGEs, but also that this tactic's effectiveness was limited by competition with other MGEs coinfecting the same cell. A further observed behaviour we



Abbreviations: BC, BAPS cluster; BLAT, BLAST-like alignment tool; CDS, coding sequence; CRISPR, clustered regularly interspaced short palindromic repeats; dsDNA, double-stranded DNA; GI, genomic island; HDT, horizontal DNA transfer; ICE, integrative and conjugative elements; I cell, infected cell; MGE, mobile genetic element; MH, more horizontal MGE; MI, intermediate MGE; MV, more vertical MGE; PRCI, phage-related chromosomal island; S cell, susceptible cell; ssDNA, single-stranded DNA. hypothesised to reduce elimination by transformation was MGE activation when cells become competent. Our model predicted that this response was effective at counteracting transformation independently of competing MGEs. Therefore, this framework is able to explain both common properties of MGEs, and the seemingly paradoxical bacterial behaviours of transformation and cell–cell killing within clonally related populations, as the consequences of intragenomic conflict between self-replicating chromosomes and parasitic MGEs. The antagonistic nature of the different mechanisms of HDT over short timescales means their contribution to bacterial evolution is likely to be substantially greater than previously appreciated.

#### Author Summary

Bacteria are able to rapidly change their characteristics, such as antibiotic resistance, by acquiring genes from surrounding cells. Some sets of genes, called mobile genetic elements (MGEs), drive their own movement between bacterial cells by transmitting in viral or viral-like manners. These selfish genes spread themselves, even if harmful to their bacterial host. Some bacteria also actively absorb DNA from the environment, a process known as transformation. Why bacteria employ transformation is controversial. Importing all DNA molecules at the same rate means cells acquire detrimental and beneficial mutations at equal frequencies. However, transformation does exhibit a known bias: it preferentially imports shorter DNA molecules, meaning it tends to delete, rather than insert, genes. By incorporating this bias into computer simulations, we show that the spread of a selfish mobile genetic element between bacteria is inhibited when cells use transformation to delete it from their genomes. We hypothesised that this ability to delete selfish mobile genetic elements is an important function of transformation. To test this hypothesis, we analysed longitudinal bacterial samples from individuals colonised by S. pneumoniae and found the bacterial cells often differed in the mobile genes they contained while still in the same host and are, therefore, able to exchange DNA by transformation. Additionally, transformable bacteria had fewer selfish mobile genes than related nontransformable bacteria. We also identified examples of mobile elements that employ various tactics to prevent their host cells undergoing transformation, indicating that some selfish mobile genes are able to circumvent deletion-by-transformation. Hence, we conclude bacterial evolution is strongly influenced by an ongoing arms race between cells and selfish mobile genes.

Horizontal exchange of DNA is common in many bacterial species [1,2], and it has become clinically relevant in recent decades through facilitating the spread of antimicrobial resistance [3] and evasion of vaccine-induced immunity [4]. However, the horizontal movement of sequence between cells is an ancient process, as revealed by its substantial effects on the overall tree of life [5]. This is despite the many risks to a recipient cell from the acquisition of DNA from an external source, such as the replicative, transcriptional, and metabolic burden of new genes, as well as the possible disruption of regulatory and protein interaction networks [6]. Perhaps most importantly, there is the potential for the acquisition of genomic parasites, against which all genomes of self-replicating cells must defend themselves [7].

Horizontal DNA transfer (HDT), a term we define to encompass all movement of heritable genetic material whether or not it alters the recipient genome, is frequently driven by such

parasitic loci, grouped together as mobile genetic elements (MGEs). MGEs encode at least some of the machinery required for their transfer between cells (horizontal transmission), and consequently drive two of the three principal mechanisms of HDT in bacteria: conjugation, the movement of DNA through an MGE-encoded conjugative pilus [8]; and transduction, the movement of DNA through an MGE-encoded virion particle [9]. To defend themselves, cells encode means of inhibiting this lateral spread, such as restriction-modification and clustered, regularly interspaced, short palindromic repeats (CRISPR) systems [10,11]. However, many MGEs insert into their host cell's chromosome, and post-integration, these defences cannot prevent such genomic parasites from subsequently maintaining a stable association with their host and passing into descendants (vertical transmission).

The third principal mechanism of HDT is transformation, the import of exogenous DNA that can be incorporated into the genome through homologous recombination, first identified in S. pneumoniae [12,13]. Natural genetic transformation is not driven by MGEs, but instead facilitated by competence machinery encoded by the bacterium itself; while not ubiquitous across bacteria, the competence machinery is usually conserved across a species [14,15]. Typically, the first step of transformation involves binding of double-stranded DNA (dsDNA) to a surface receptor (ComEA in gram-positive bacteria; ComE in gram-negative bacteria). This requires a pseudopilus formed of ComY or ComG proteins (in gram-positive bacteria) or Pil proteins (in gram-negative bacteria) [14,16,17]. The bound DNA then passes through a specialised pore (ComEC in gram-positive bacteria; ComA in gram-negative bacteria) [18] that translocates the nucleic acid into the cytosol in a single stranded form, with the concomitant degradation of the complementary strand [19,20]. In gram-positive bacteria, ComFA appears to have a role in driving this DNA import [21]; in the gram-negative bacterium Haemophilus influenzae, ComM has been identified as playing a role in transformation post-import [22]. Upon import, this single-stranded DNA (ssDNA) is itself cut into fragments [23,24], with a median length of approximately 6.6 kb in S. pneumoniae [23]. These ssDNA fragments are then bound by proteins inside the cell, culminating in the formation of a RecA nucleoprotein filament [25,26]. This is the form in which the ssDNA can invade the host chromosome duplex, potentially resulting in homologous recombination [27]. The components of this highly-specialised competence machinery are encoded by multiple nonmobile loci within the cellular genome.

With the apparent exceptions of *Helicobacter pylori* and the Neisseriaceae, naturally transformable bacteria tend to tightly control expression of the competence machinery [15], making it difficult to identify the full range of species in which the system is active [28,29]. Distinct quorum-sensing systems based on secreted peptide pheromones are used in S. pneumoniae [30], Bacillus subtilis [31] and Streptococcus thermophilus [32], whereas nonpeptide autoinducers regulate expression in *Vibrio cholerae* [33]. These signals are then transduced to a regulator of transcription, such as the competence-specific sigma factor  $\sigma^{X}$  in S. pneumoniae [34] or the ComK transcriptional regulator in *B. subtilis* [35]. In addition to the genes directly required for competence, these often activate a range of other coordinated activities as part of what appears to be a broader stress response, referred to by terms such as 'K state' or 'X state' [15]. Notably, in the well-studied species S. pneumoniae and B. subtilis, competence has been found to be coordinated with the secretion of bacteriocins that kill noncompetent cells in a process termed "fratricide" or "cannibalism" [36], as well as cell cycle arrest [37,38]. In B. subtilis, the noisy transcription of *comK* results in phenotypic differentiation termed "bet hedging" [39,40], with the population reaching a dynamic equilibrium in which a subset of the population actively replicates, while around 15% of the population are competent, nonreplicative, persister cells [41,42]. Hence, competence is expressed in a diverse set of patterns in different species.

What selective advantage transformation provides to the cell remains controversial [15,43]. Three primary groups of hypotheses have been proposed as explanations. The first is that HDT

facilitates the acquisition of beneficial genetic polymorphisms and, therefore, can be considered analogous to eukaryotic sexual reproduction. Straightforward forms of these models can demonstrate that recombining populations often have a higher fitness than nonrecombining populations [44-46], but such explanations are disfavoured, as they rely upon group selection [47]. Identifying an advantage at the level of the individual is difficult, as the competence system facilitates the acquisition of beneficial and deleterious sequence at the same rate in these models. One explanation is the existence of synergistic epistasis, the situation in which polymorphisms interact such that their cumulative effect on an individual's fitness is greater than the sum of their individual effects [48-50]; however, experimental investigation of adaptive changes from in vitro evolution experiments has found antagonistic epistasis between mutations to be more common [51,52]. Another set of models uses the scenario of recombining cells entering a new niche [53], such that acquired alleles are biased toward being beneficial, as they arise from donors already better adapted to the new conditions. Transformation can, therefore, be advantageous if competent cells encounter different environments continuously [46] or periodically [41], akin to the "Red Queen" hypothesis that individuals must unceasingly change to avoid decreasing in fitness [54]. However, exchanges of DNA between different lineages of transformable bacteria are not frequent enough to disrupt predominately clonal population structures [55-57], with years often elapsing between substantial imports of divergent sequence [58].

The infrequent exchange of DNA between lineages does not preclude exchange of sequence between closely related genotypes facilitating the repair of deleterious mutations. Owing to the physical structuring of bacterial populations, it is much more likely that cells will undergo recombination with clonally related neighbours than with other, divergent lineages. Nevertheless, such a mechanism seems unlikely to be the primary purpose of the competence machinery. Transformation cannot distinguish recent point mutations from the alleles that repair such spontaneous changes; furthermore, previous models have identified the problem that if deleterious mutations are frequent and often result in cell lysis, then the pool of DNA available for transformation will be enriched for lower fitness alleles [48]. Additionally, transformation events that reverse the most commonly observed mutations in *S. pneumoniae* are efficiently inhibited by the mismatch repair system [59,60]. Effective against particular base substitutions but not large insertions or deletions, this repair system also has lower, but detectable, activity in *B. subtilis* [61,62] and *H. influenzae* [63] and is present in many transformable species [64].

The second set of hypotheses suggest that imported DNA is used as a template to repair dsDNA breaks [65]. Potential mutagens have been found to increase the rate of transformation in S. pneumoniae [66], Legionella pneumophila [67], and H. pylori [68], with conflicting data as to whether the same regulation is observed in *B. subtilis* [69–72]. However, experimental work detected no such regulation in S. thermophilus [73] or H. influenzae [71,74]. Similarly, there is some evidence for transformation increasing resistance to ultraviolet exposure from experimental work on B. subtilis [70], but the same result was not observed in H. influenzae [74,75] or L. pneumophila [67]. While all bacteria suffer dsDNA breaks as part of normal DNA replication, transformation is found in distantly related species that are unlikely to be subject to particularly high or variable burdens of DNA damage, such as the nasopharyngeal commensals S. pneumoniae, H. influenzae, and Neisseria meningitidis. Correspondingly, the SOS response of H. influenzae lacks the translesion repair system [76], important in repair of mutagen-induced damage in *B. subtilis* and many other species, while both N. meningitidis and S. pneumoniae generally lack identifiable SOS responses [64,77] despite the import of translesion repair genes into at least one S. pneumo*niae* isolate on a conjugative element [78]. Hence, there is no strong evidence that the competence system functions as an alternative to, or an enhancement of, the known role of the SOS response in ameliorating the effects of mutagens, although comparisons between the gene content of species in different niches are confounded by the associated variation in genome size [79].

The third set of hypotheses are based on the competence system functioning as a means of scavenging nucleotides from the environment [43,80]. This is consistent with only the minority of transforming DNA being integrated into the chromosome [19], as well as competence being induced by purine starvation in *H. influenzae* [81] and by nucleoside starvation in *V. cholerae* [82]. Although not known to be naturally transformable itself, orthologues of competence genes in *Escherichia coli* have been found to facilitate nutrient acquisition in vitro [83]. However, some species only develop competence in the absence of nutrient starvation, and others limit the molecules they import in a sequence-specific manner to avoid acquiring DNA from other species [15]. Additionally, it is difficult to understand how the import of a single DNA strand as a protected nucleoprotein filament, with the other strand degraded extracellularly, maximizes the acquisition of nutrients, whereas it is optimised for integration of imported DNA into the chromosome [41].

Here, we present a novel hypothesis motivated by several key characteristics of HDT by transformation. The first is that physical structuring of populations means most exchanges are between isogenic, or near-isogenic, cells. The second is that in vitro and in vivo characterisation of recombinant *S. pneumoniae* [4,60,84] and *H. influenzae* [85] isolates has demonstrated that transformation results in the integration of DNA tracts with an approximately geometric length distribution. The majority of such recombinations have been shown to be shorter than common MGEs [60,85,86], which are typically several kilobases in length and often much larger [87]. The third is that homologous recombination is able to span regions of dissimilar sequence in either imported or chromosomal DNA, meaning transformation can alter genome content [88,89]. As the acquisition of longer DNA molecules is limited by extracellular degradation, cleavage on import [23,24], and restriction endonuclease cleavage of novel sequence [89], transformation has a tendency to delete, rather than import, genes. Hence, if infected and uninfected cells exist in close proximity, transformation should act to remove MGEs far more efficiently than it spreads them, making it an effective means of inhibiting the vertical transmission of genomic parasites.

#### Results

#### Asymmetrical Transfer Benefits Transformable Cells

To explore potential benefits of different evolutionary strategies to bacterial cells, a stochastic compartmental model was developed to simulate genetic exchange through HDT (see Methods). The first simulations investigated scenarios in which transformation might facilitate the import of beneficial foreign DNA, corresponding to either the acquisition of novel loci or the reversal of Muller's ratchet by repairing deleterious mutations. These featured two strains of different fitnesses, one of which was competent for transformation, growing and competing in a homogeneous environment. Increasing the rate of transformation,  $\tau$ , was advantageous to the transformable strain so long as the nontransformable strain was fitter and, therefore, donating beneficial alleles (Fig 1A). The greater the fitness advantage of the donor, the greater  $\tau$  needed to be for the transformable strain to acquire the beneficial allele before it was outcompeted. However, when the nontransformable strain was less fit, increasing  $\tau$  was detrimental to the transformable strain, which could only diversify through the acquisition of deleterious alleles. This situation was exacerbated if transformation has an associated fitness cost, set at 5% in Fig 1B. In these simulations, the cost of expressing the competence machinery meant the transformable strain was outcompeted by a fitter competitor more quickly, limiting the opportunity for the acquisition of beneficial loci, and outcompeted a less fit competitor more slowly, increasing the opportunity for the acquisition of deleterious alleles. Hence, in these simulations, transformation's ability to facilitate exchange between strains can only provide a long-



Fig 1. Variation in the effectiveness of transformation when importing DNA from a different strain. (A) Heatmap showing the outcome of simulated competitions between two strains, of which only one is transformable. Each cell in the grid represents a specific transformation rate ( $\tau$ ) and relative fitness of the allele at an exchangeable locus within the transformable strain, displayed at the top of each column. Relative fitnesses greater than one indicate the transformable strain has an initial advantage over the nontransformable strain; relative fitnesses below one indicate the nontransformable strain has the initial fitness advantage. Each cell is split in two: the "S" component shows the outcome of simulations in which transformation is symmetrical, and the "A" component shows the outcome of simulations in which transformation is asymmetrical, with the lower fitness allele acquired at a rate 10-fold lower than that of the higher fitness allele. (B) Heatmap showing the outcome of simulated competitions between two strains, of which only one is transformable, when there is a cost associated with the expression of the competence machinery. This figure shows the outcome of simulations analogous to those displayed in panel A, except that the transformable strain has a growth rate,  $\gamma$ , 5% lower than the nontransformable strain to represent the cost of the competence machinery. This cost was constant across simulations and independent of the relative fitness difference between the alleles of the exchangeable locus. Raw data are tabulated in <u>S1 Data</u>.

doi:10.1371/journal.pbio.1002394.g001

PLOS

BIOLOGY

term benefit to cells if they continuously encounter potential donors of alleles that increase their fitness to an extent that outweighs the cost of expressing the competence machinery, and those beneficial alleles can be acquired more quickly than the transformable cells are outcompeted by the potential donors.

The symmetrical shuffling of alleles in this scenario only occurs if donor and recipient have similarly sized alleles of a shared locus. However, homologous recombination is able to transfer DNA when there is similar sequence only at both ends of an otherwise divergent locus, even if the intervening sequence is the length of a typical genomic island [60]. Homologous recombination can, therefore, change a genome's content through integrating or deleting components of the accessory genome, depending on whether the intervening locus is present in the imported DNA or recipient genome, respectively (Fig 2A). If such allelic variation in the length of a locus exists, then all else being equal, transformation has a bias towards shorter alleles, resulting in a tendency to delete, rather than import, sequence [90–93]. This is a consequence of the necessity for similarity at each end of an imported DNA fragment for it to be stably integrated into a chromosome; any cleavage that separates one of these regions from the other, which becomes increasingly likely as the distance between the homologous arms lengthens, prevents the integration of the entire intervening imported sequence [94].

# PLOS BIOLOGY



**Fig 2.** Modelling the interaction of HDT mechanisms in a simple bacterial population. (A) Definition of the asymmetry parameter, φ. (B) Illustration of the exchange of DNA within and between clonally related cells. (C) Description of the stochastic compartmental model of within-population HDT.

doi:10.1371/journal.pbio.1002394.g002

The parameter  $\varphi$  is included in our model to represent the asymmetric integration of alleles of different lengths (model parameters are summarised in S1 Table). When  $\varphi = 1$ , both alleles of such loci are exchanged at an equal rate (symmetrical transformation); if  $\varphi < 1$ , then the shorter allele is imported at a higher rate than the longer allele, a situation reversed if  $\varphi > 1$ . Assuming imported DNA fragments have a geometric length distribution with a rate parameter of  $\lambda_{\rm R}$  bp<sup>-1</sup> [60], then  $\varphi$  is a function of an insertion's length, x, of the form  $\varphi_x = (1 - \lambda_{\rm R})^x$ . The simulations were run as previously described, with a transformable and nontransformable strain initially distinguished by a biallelic locus, with one strain having a neutral short allele and the other a long allele associated with a fitness cost. The transformable cells acquired long alleles at a rate of  $\varphi = 0.1$  relative to the short allele (see <u>Methods</u>). In simulations in which the deleterious insertion was initially present in the transformable strain, transformation was beneficial in facilitating removal of the locus. However, increasing rates of transformation were neutral when the transformable strain was fitter, as any rise in the rate at which the longer allele was acquired was compensated for by its more rapid removal, a pattern observed whether or not there was a fitness cost associated with the expression of the competence machinery (Fig 1A and 1B; S1 Data). Hence, asymmetric transformation can purge genomes of deleterious insertions.

#### Asymmetrical Transformation Counteracts MGE Spread

In these simulations, the deleterious long alleles were eventually removed from the population by transformation; following their elimination, there is no longer a benefit to retaining the competence system. However, deleterious insertions are continuously generated within bacterial populations through MGE movement [57]. We extended the stochastic compartmental model of bacterial HDT such that the deleterious MGEs transmitted horizontally within a community of cells at a rate parameterised by  $\beta$ , assumed to be much faster than the rate of MGE transmission between communities ( $\beta_i$ ; Fig 2B and 2C). The model simulated the growth, competition, and DNA exchange between susceptible (S) and infected (I) bacteria, isogenic except for a single locus at which an MGE was present in the I bacteria. Transformation was only able to eliminate MGEs when they were integrated into the host chromosome. MGEs excised themselves from the host chromosome at rate *f*, which determined the rate at which they transitioned from vertical to horizontal transmission. Simulations were conducted with two MGE types: "more horizontal" (MH;  $\beta = 10^{-6}$  unless stated, b = 10, f = 0.05,  $c_M = 0.075$ , a = 1), which frequently transmitted between cells, and "more vertical" (MV;  $\beta = 10^{-3}$  unless stated, b = 5, f = 0.005,  $c_M = 0.0025$ , a = 0), which more stably associated with host chromosomes.

The first set of simulations (Fig 3A) compared different  $\varphi$  values against MGEs of varying transmissibility, with a fixed transformation rate of  $\tau = 10^{-4}$ . We found that simulations with  $\varphi > 1$ , which favoured the transfer of insertions, actively drove the spread of MGEs through the cell population. However, when asymmetry favoured the import of shorter alleles ( $\varphi < 1$ ), transformation was highly effective at inhibiting the spread of MH MGEs over a narrow range  $\beta$  values, and MV MGEs across all tested  $\beta$  values. Similarly, when  $\varphi = 0.1$ , higher rates of  $\tau$  were found to be more effective against MV across all tested values of  $\beta$ , whereas MH could still spread through the population if sufficiently transmissible (Fig 3B). Sensitivity analyses indicated elimination of MGEs was dependent upon a sufficiently high concentration of extracellular DNA for homologous recombination, facilitated by higher cell growth rates and greater stability of extracellular molecules (S1 Fig).

This mechanism was also found to provide a benefit at the level of the population. Independent simulations were run in which cells were either nontransformable or transformable ( $\varphi = 0.1, \tau = 10^{-4}$ ) and, therefore, affected by an associated cost of expressing the competence



**Fig 3.** Identifying the necessary conditions for constitutive competence to be effective in inhibiting MGE transmission. (A) Heatmap showing the outcome of simulations in which MGEs infect cells competent for transformation at rate  $\tau = 10^{-4}$ . The colour of the heatmap represents the proportion of the population infected by MGEs over the course of each simulation. Each cell represents a specific MGE transmissibility ( $\beta$ ) and transformation asymmetry ( $\phi$ ); the "MH" and "MV" components show simulations with MGEs having relatively greater propensities for horizontal and vertical transmission, respectively. (**B**) Heatmap, displayed as in panel A, but this time comparing the effects of varying  $\beta$  against changing the rate of transformation ( $\tau$ ) with a fixed value of  $\phi = 0.1$ . (**C**) Conditions necessary for transformation to provide a fitness advantage to the population. This heatmap shows the ratio of the total number of cells in two independent sets of simulations: one in which cells were nontransformable, and another in which the cells were transformable and suffered an associated cost ( $c_c$ ). In both sets of simulations, either the MH or MV MGEs were present in the populations, each of which was associated with a varying cost to the host ( $c_m$ ); transformation was only effective at inhibiting the transmission of MV. The higher the value of the ratio, indicated by the heatmap colour, the relatively greater the size of the bacterial population when cells were transformable. Raw data are tabulated in <u>S1 Data</u>.

doi:10.1371/journal.pbio.1002394.g003

machinery ( $c_C$ ). Each population featured either MH or MV, with the associated cost of being infected by such an MGE,  $c_M$ , varying between simulations. The heatmap summarizing the ratio of the total cell population recorded from the simulations with identical parameterization, but differing in whether or not the competence machinery was expressed, is shown in Fig <u>3C</u>. In simulations involving MH, against which transformation was ineffective, either the total number of cells was similar in the matched sets of simulations, if  $c_C$  was negligible, or the nontransformable cells were detectably more successful, if  $c_C$  was high. However, transformation was effective at inhibiting the spread of MV, and, therefore, in simulations involving this MGE in which  $c_M$  was greater than  $c_C$ , transformation resulted in the cell population being more numerous when transformable, despite the associated costs of the competence machinery. Hence, constitutive asymmetric transformation inhibits the vertical transmission of deleterious MGEs, thereby allowing bacteria to purge such parasites from their genomes, potentially resulting in increased fitness of individual cells and populations.

#### Transient Competence Removes Deleterious MGEs

In many species, competence is only transiently expressed, rather than being constitutive. To test whether these expression patterns are compatible with inhibiting the spread of MGEs, populations were simulated in which cells were only competent in a "C state." As competence is often regulated by diffusible signals, cells were assumed to enter C state upon an intercellular signal, a "C signal" ( $s_C$ ) constitutively generated by all cells, surpassing a threshold,  $t_C$ . C-state cells suffered a cost to expressing competence, quantified as a growth inhibition  $c_C$ , and exited C state at the rate  $r_C$  (Fig 4A). These parameters were sufficient to define a cell-density-dependent C state ( $g_C = 10$ ,  $r_C = 0$ ,  $c_C = 0.1$ ), in which cells became irreversibly competent above a particular density threshold, and a "bet hedging" strategy ( $g_C = 0.1$ ,  $r_C = 0.5$ ,  $c_C = 1$ ), which resulted in a dynamic equilibrium in which approximately 13% of the population was in C state, with their growth completely arrested, at any point in time. Neither caused much change in the overall growth pattern of the population (Fig 4B).

C state can also mimic "fratricide" or "cannibalism" [15]. Lysis of surrounding bacteria, which are usually near-isogenic clonal descendants of a recent common ancestor, may be important in raising the concentration of DNA available for transformation. To explore the biological significance of this, C-state cells were enabled to kill non-C-state cells at a rate  $k_{\rm C}$  (Fig 4A). This inclusion of transient arrest of cell growth and cell–cell killing ( $c_{\rm C} = 1$ ,  $k_{\rm C} > 0$ ) resulted in oscillatory growth patterns in which populations went through alternate phases of clonal growth and competence. At  $k_{\rm C} = 10^{-6}$ , competence was transient but associated with little change in population size, while  $k_{\rm C} = 10^{-3}$  drove large population oscillations. Increasing  $r_{\rm C}$  had the opposite effect on oscillation amplitude and frequency (Fig 4B). Large, unexplained population oscillations have been observed during the growth of *S. pneumoniae* in a chemostat in the absence of phage [95,96], and their occurrence during carriage could explain the discrepancy between census and effective population sizes in animal models [97].

Like constitutive competence, transient competence was also effective in inhibiting the spread of MV (Fig 4C). Irreversible cell-density-dependent C state eliminated MV from the population more quickly than bet hedging, although the latter strategy confined the costs of expressing competence to only a subset of the population. Oscillatory growth patterns were effective when transformation was rapid, necessitating a high value of  $\tau$  and sufficient DNA release to sustain transformation throughout the competence period. This was facilitated by the cycles of cell–cell killing and rapid growth driven by  $k_c > 0$ , and a shorter period during which competence-associated arrest of growth applied (S2 Fig). By contrast, the spread of MH was only inhibited by large oscillations in cell population size (Fig 4C), also driven by higher values



**Fig 4. Inhibition of MGE transmission by transient competence.** (**A**) Model of a transient competent ("C") state and cell–cell killing of non-C-state cells. (**B**) Graph showing the original cell growth curve (no C state), cells entering the C state in a cell-density-dependent manner, then never leaving ( $k_c = 0$  and  $r_c = 0$ ); a "bet hedging" strategy ( $k_c = 0$ ,  $g_c = 0.1$ ,  $r_c = 0.5$ ) in which only a fraction of the population is competent for transformation at any one time; cells undergoing small population oscillations ( $k_c = 10^{-6}$ ) and cells undergoing large population oscillations ( $k_c = 10^{-3}$ ) at frequencies determined by  $r_c$ . (**C**) Heatmap summarising the outcomes of simulations comparing patterns of growth and competence expression in panel B with different MGEs. Colours are scaled as in Fig 3. Results for two representatives of MH and MV are shown, each associated with different rates of HDT. Transformation at the specified  $\tau_c$  rate occurred in the C state, such that the cells that never entered the C state were never competent for transformation. Raw data are tabulated in <u>S1 Data</u>.

doi:10.1371/journal.pbio.1002394.g004

of  $k_{\rm C}$ , although this effect was contingent upon the MGEs being relatively unstable outside the cell, such that they could not survive extracellularly from one population boom to the next (S3 Fig). Hence, we find that coregulation of cell–cell killing with asymmetric transformation, as modelled by this C state, can synergistically combine to block the transmission and spread of MGEs. Horizontal transmission is limited by the large reductions in the density of susceptible cells during population crashes, which are contemporaneous with large releases of DNA that can limit vertical transmission through transformation. Importantly, all simulated patterns of growth and competence were found to be compatible with the elimination of parasitic MGEs.

Some MGEs have spread successfully while carrying cargo, such as antibiotic resistance, that can be beneficial to their bacterial host [4,84]. To explore the spread of beneficial MGEs, we allow

fitness "costs" to be negative, corresponding to advantages that increase the replication rate. Fig 5 shows the spread of beneficial and deleterious MV when the cell population undergoes irreversible density-dependent competence (Fig 5A), low-amplitude oscillations (Fig 5B), or high-amplitude oscillations (Fig 5C). Constitutive, density-dependent transformation is the most effective at eliminating MV, regardless of whether they are detrimental or beneficial. By contrast, transient competence is able to remove deleterious MV but permit those that are beneficial to the cell to spread over multiple orders of magnitude variation in  $\tau$ . This is the consequence of the MGEs increasing in frequency between phases of competence through both transmission and selection. However, at the population level, asymmetric transformation is still required to preferentially acquire beneficial alleles, with tightly regulated expression of competence alone being insufficient (S3 Fig). Hence, transiently competent populations biased against acquiring insertions through transformation can often allow beneficial MGEs to spread, while eliminating deleterious MGEs.

#### Frequent Opportunities for MGE Removal by Transformation

For cells to derive a benefit from transformation acting to remove MGEs, there must be frequent instances of deleterious MGEs being polymorphic within otherwise isogenic cell populations. We explored the potential for such situations within a dataset of over 3,000 sequenced isolates of *S. pneumoniae*, of which 1,715 represented longitudinal sampling of 371 hosts [98]. By performing a systematic search for variation in the three major classes of MGEs found in the pneumococcus (phage, integrative conjugative elements, and phage-related chromosomal islands [PRCI] [57]), we uncovered multiple instances of MGE variation between closely related isolates from the same host sampled on different days, despite the comparative insensitivity of sampling a single colony per timepoint. All but one of the well-characterised examples involved changes in prophages, which integrate into the genome and do not typically contain beneficial cargo genes in *S. pneumoniae* [57].

Phylogenies were constructed that accounted for divergence through interstrain transformation, thereby reconstructing clonal descent (S4 Fig) [99]. Such analysis of the most common lineage in the sampled population, BAPS cluster (BC) 1-19F, demonstrated that the MGE variation within carriage represented changes in the prophage content of stably carried bacteria with very similar core genomes, rather than a host acquiring new genotypes with stable MGE content (Fig 6, S2 and S3 Tables). The detectable frequency of integrative MGE acquisition indicates these otherwise isogenic, recombining populations will frequently consist of coexisting S and I cells. That this coexistence may persist over weeks or longer is suggested by the repeated identification of the same MGE within a particular host, contrasting with its absence from other timepoints during individual carriage episodes. However, many bacteria were nonlysogenic, and there is little evidence of prophages being conserved over substantial proportions of the lineage's overall evolutionary history, consistent with selection against cells infected with these parasites (S5 Fig) [57,100,101].

The precise mechanisms by which the MGEs may be eliminated are difficult to differentiate in populations with few distinguishing genetic markers. However, in one of the rare cases in which interstrain DNA transfer was detected within a single carriage episode, a transformation event was associated with the loss of an otherwise stable PRCI (<u>S6 Fig</u>) [57]. This observation of an MGE apparently being removed by transformation demonstrates the feasibility of the mechanism underlying our model.

#### Recurrent Inhibition of Transformation by Prophage

The leaf nodes of the phylogenies shown in <u>Fig 6</u> are marked according to the status of their *comYC* gene, required for effective transformation [4,15]. <u>S5 Fig</u> shows that all these instances





Fig 5. The effect of different patterns of competence expression on the transmission of MGEs that benefit their host. (A) Heatmap summarising the outcome of simulations in which MV-type MGEs infected cells that entered the C state in a cell-density-dependent manner ( $r_c = 0$ ). Each cell represents a specific transformation rate  $\tau_c$  and "cost" of the MGE ( $c_M$ ); negative costs imply the MGE benefits the cell. Each cell is split into two components, representing an MV MGE with high ( $\beta = 10^{-1}$ ) or low ( $10^{-3}$ ) transmissibility. (B,C) Heatmaps, displayed as in panel A, but comparing cells transiently entering the C state with  $k_c = 10^{-6}$  (panel B) or  $k_c = 10^{-3}$  (panel C;  $r_c = 0.9$  and  $\phi = 0.5$  in both cases). Compared to cell-density-dependent C state, these patterns of C state expression were still effective at inhibiting the spread of MV that were detrimental to the cell, but were a relatively lower impediment to the spread of MV that benefitted the cell. Raw data are tabulated in <u>S1 Data</u>.

doi:10.1371/journal.pbio.1002394.g005





**Fig 6. Evidence for coexistence of susceptible and infected BC1-19F cells within individual carriage episodes.** The displayed phylogeny was generated based on point mutations, excluding base substitutions likely to have been introduced by recombination (S4 Fig). Leaf nodes are annotated to indicate whether the *comYC* gene, required for efficient transformation, is intact (green dash) or disrupted by a prophage insertion (orange dash; see key in Figs 8 and S5). Specific cases of changes in prophage content within what are likely to represent individual carriage episodes are highlighted. Each box displays the annotation of a specific prophage (S2 Table), along with sequence read mapping heatmaps beneath showing the depth of coverage across

the viral sequence for individual isolates from a single host: blue for low levels of mapping, indicating the sequence is absent, and red for high levels of mapping, indicating it is present (see scale in Fig 8). The isolates are ordered by the date of isolation. Epidemiological data are summarised in S3 Table.

#### doi:10.1371/journal.pbio.1002394.g006

of *comYC* disruption result from the insertion of a prophage into the coding sequence (CDS). Whereas the integrases of most MGEs target them into the small noncoding fraction of the bacterial genome, thereby minimising the selective cost imposed on the host, this was the only example of an MGE disrupting a CDS observed in genomic data from a *S. pneumoniae* population [57]. Past investigation of resistant *S. pneumoniae* have found a second instance of a genomic island disrupting a CDS; a gene cassette encoding a *mefA* macrolide resistance determinant was observed to inhibit transformation through insertion into *comEC* [102], which encodes a protein critical for forming the DNA import pore.

The benefit to the MGE of disrupting the competence system of the host cell is illustrated in Fig 7A and 7B. These display repeats of the simulations shown in Fig 3A and 3B, except that infection by either MH or MV prevented their host undergoing transformation. The results indicate MGEs derive an advantage from such abrogation of host competence when the transformation rate was sufficiently high ( $\tau > 10^{-7}$ ), and asymmetry in favour of shorter alleles sufficiently strong ( $\phi \le 0.1$ ), to impede their transmission through the population. Hence, in this model, transformation's potential to prevent the spread of MGEs provides the selection pressure for such mobile elements to target competence genes for disruption when they integrate into the genome.

A second lineage from the same population, BC4-6B, included clade A that diversified similarly to BC1-19F, with intermittent import of diversity through transformation and disruption of *comYC* (Fig.8). Yet clade B shows little evidence of diversification through transformation or MGE variation (S7 and S8 Figs). This reduction in all HDT mechanisms is associated with the stable inheritance of two prophages within clade B, one of which disrupts *comYC* (S8 Fig),



Fig 7. The benefit to MGEs of inhibiting rapid, asymmetric transformation. Panels A and B repeat the simulations displayed in Fig 3A and 3B, respectively, with the difference that the MH and MV MGEs in these simulations disrupt the host's ability to undergo transformation when they insert into the cell's chromosome. Raw data are tabulated in S1 Data.

doi:10.1371/journal.pbio.1002394.g007



Leaf node label key
- Intact *comYC* gene



doi:10.1371/journal.pbio.1002394.g008

BIOLOGY

PLOS

exemplifying the efficient vertical transmission of prophages in the absence of transformation. However, as is necessary for competence to be preserved in the species [74], most insertions into *comYC* were not successful. Instead, across BC1-19F and clade A of BC4-6B, noncompetent bacteria appear to have been removed by selection at a faster rate than diversification through interstrain transformation, with "clonal" lineages only rarely found to transmit between multiple hosts.

#### MGEs Inhibit Transformation in Many Species

The insertion of a prophage into *comYC* is not restricted to *S. pneumoniae*. Searches for orthologues of the relevant phages' integrases, which determine the site into which a phage inserts, identified examples of a prophage targeting the same gene in several other related species; these included *Streptococcus mutans* and *Streptococcus parauberis*, from the same genus, as well as an example in *Lactococcus lactis* (Figs 9A and S9). These all inserted at an orthologous, but not perfectly conserved, site within the gene (S10 Fig). Unexpectedly, an orthologous integrase was found in *Streptococcus agalactiae*, a species not considered to be naturally competent (Fig 9B) [28,103]. This integrase was part of a phage that inserted into the *cas3* gene of *S. agalactiae*'s CRISPR2 locus [104]. While disruption of loci that inhibit infection of the cell would not be expected to benefit a prophage, CRISPR systems are capable of targeting integrated prophages, resulting in cell suicide, or the post-activation excised, replicating form of the phage [105–107]. Under either scenario, depleting the cytosol of functional CRISPR proteins would provide the phage with an advantage prior to transmitting horizontally to the next cell. Hence, cellular countermeasures to both defences against horizontal and vertical transmission of MGEs are targeted by similar integrases directing the insertion of prophages in streptococci and related genera.

A different example was observed in *Streptococcus equi* [108], in which a prophage inserted into the comFA gene. Orthologues of this distinct integrase were identified in other species including the zoonotic pathogen Streptococcus suis, in which the protein again directed a prophage to insert into the host cell's comFA gene (Fig 9C). Similarly, some strains of Listeria monocytogenes have a prophage inserted into their comK genes [109], encoding a regulator of competence. A similar insertion was identified in a representative of *Listeria innocua* [110], a further example of which is shown in <u>S11 Fig.</u> Searching for proteins similar to the *L. monocy*togenes prophage integrase identified a prophage inserted into comFA in representatives of the Bacillus cereus and thuringiensis group (Figs 9D and S11). However, codon alignments demonstrated that the insertion site within the Bacillus comFA genes was distant from that of the distinct prophage identified in S. suis (S10 Fig), suggesting the targeting of this gene represented convergent evolution between the two MGEs. Another orthologue of the integrase found in L. monocytogenes was present in a phage of Enterococcus faecalis, this time targeting the MGE to insert into radC (S11 Fig); the same gene was reported to be targeted by a prophage inserted into *B. subtilis* that inhibited transformation [111], although the insertion site was not demonstrated to be the cause of this inability to integrate exogenous DNA. While RadC levels increase during competence, it does not always appear to be essential for transformation in the laboratory [<u>112</u>].

Another previously identified example of MGEs inhibiting competence was the observation from *Aggregatibacter actinomycetemcomitans* genomes that some MGEs inserted into *comM* [113], which encodes a protein important for efficient incorporation of DNA into the chromosome through homologous recombination [22]. MGEs were also inserted into *comM* in a single representative of *Acinetobacter baumannii* [114], and a large insertion was identified in the same gene in *Mannheimia succiniciproducens* [115]. Searching for orthologues of the integrase targeting *comM* from *A. actinomycetemcomitans* identified examples in other strains of *A*.

#### 



**Fig 9. Selected examples of MGE insertions disrupting chromosomal protein coding sequences.** (A) Comparison between *Lactococcus lactis* isolates IL1403 and KLDS 4.0325, the latter of which has a prophage inserted into the *comYC* gene, encoding the major structural component of the competence pilus. The sequences' accession codes are given in brackets underneath the isolate names. Blue and orange boxes represent CDSs, with the direction of their transcription indicated by their vertical position relative to the horizontal line; pink boxes indicate putative MGE CDSs in the same way. Brown boxes linked by dashed lines mark the fragments of a pseudogene disrupted by MGE insertion. The red bands link regions of similar sequence in the two loci, as identified by BLAST-like alignment tool (BLAT); the intensity of the colour indicates the strength of the match. The prophage integrase has ~51% identity with the protein that drives integration into the orthologous gene in *S. pneumoniae* 670-6B (SP670\_2190). (**B**) Comparison between *Streptococcus agalactiae* isolates COH1 and FSL S3-277, the latter of which has a prophage inserted into the *camYC* gene of *S. pneumoniae* 670-6B. (**C**) Comparison of *Streptococcus suis* isolates P1/7 and 89–590, the latter of which has a prophage inserted into the *3* half of the *comFA* competence gene. This prophage integrase is ~46% identical with



that of the prophage inserted into the orthologous gene in *Streptococcus equi* (SEQ\_1765). (**D**) Comparison between *Bacillus cereus* isolates MHI 226 and VD214, the latter of which has a prophage inserted into the 5' half of the *comFA* competence gene. The prophage's integrase is ~44% identical with that of the prophage inserted into the *comK* competence gene in *Listeria monocytogenes* (LMRG\_01511).

doi:10.1371/journal.pbio.1002394.g009

*baumannii* (S12 Fig), and similar insertions were observed in genes encoding orthologues of ComM across a diverse set of species, including the animal pathogen *Mannheimia haemolytica*, the human pathogen *Francisella philomiragia*, and the plant pathogen *Pseudomonas syringae*.

#### The Effect of Competition between MGEs

Despite the potential advantage to individual mobile elements, the disruption of competence by MGEs is not ubiquitous. To investigate the reasons underlying this, we modelled an MGE "MI," intermediate in properties between MV and MH ( $\beta = 5 \times 10^{-6}$ , b = 7, f = 0.01,  $c_M = 0.005$ , a = 1), which is able to spread through a nontransformable cell population but is eliminated by cell-density-dependent, bet-hedging, and transient patterns of competence expression (Fig 10A). However, "MI<sub>NT</sub>," which has the same properties but disrupts the competence system of the host cell, is able to spread regardless of the type and rate of transformation of the host cell. When MI and MI<sub>NT</sub> coinfect the same cell population, both MGEs achieve similar levels of transmission, mirroring the stability of both prophages within BC4-6B clade B, despite only one inhibiting competence (S8 Fig). This limits the advantage of strategies such as disrupting *comYC*, as the affected cell cannot eliminate superinfecting MGEs, benefitting all other elements infecting the same host.

Individual MGEs can also escape the consequences of transformation by switching from vertical to horizontal transmission when a cell becomes competent. This is achieved through increasing the activation rate, *f*, to an elevated value  $f_C$  during C state. Evidence for the relevance of this mechanism is the observation that many prophages [116–118] and integrative and conjugative elements (ICEs) [119,120] excise from the chromosome in response to elevated levels of RecA, the protein required for homologous recombination. Simulating the spread of MI with low (*f* or  $f_C = 0.005$ ) or high (*f* or  $f_C = 0.5$ ) rates of activation during either clonal growth or C state shows that relying heavily on vertical transmission ( $f = f_C = 0.005$ ) makes the MGE highly susceptible to transformation (Fig 10B). By contrast, high levels of activation outside of C state (f = 0.5) results in substantial costs to the host population, with the consequent low host cell density reducing the efficiency of horizontal MGE transmission (S13 Fig). However, activating only at a high rate during C state permits stable vertical transmission when cells are not competent, while greatly reducing elimination through transformation. Nevertheless, cells' synergistic coupling of reduced cell density with transformation can still strongly inhibit the spread of MGEs adopting this optimal strategy.

To test whether both these strategies for avoiding elimination by transformation might be synergistic, the analysis of activation rates was repeated for  $MI_{NT}$ . This found that the optimal strategy for  $MI_{NT}$  was different; as the MGE could not be removed by transformation, it could achieve fixation in a population without elevated rates of activation in C state, and instead benefitted from a low rate of activation regardless of the cell's behaviour (Fig 10C). However, when both MI and  $MI_{NT}$  were allowed to compete in the same population, both operating under optimal strategies (*f* low and *f*<sub>C</sub> high for MI; *f* and *f*<sub>C</sub> low for MI<sub>NT</sub>), MI was more successful in a greater number of scenarios (Fig 10D).  $MI_{NT}$ , by contrast, was more successful only if there were large changes in population size that inhibited horizontal transmission. This was the consequence of the rapid activation of MI, following the onset of C state, allowing it to spread horizontally at a higher rate; additionally, this behaviour facilitated killing C-state cells in which  $MI_{NT}$  was also inserted but not yet activated. Hence, elevated *f*<sub>C</sub> provides an





**Fig 10.** The effectiveness of MGE strategies for reducing elimination by transformation. (A) Heatmap summarising the outcomes of simulations comparing the patterns of cellular competence shown in Fig 4C in the presence of the MGE MI (top row), which has properties intermediate between those of MH and MV. The colour of each cell represents the proportion of the population infected by the MGE over the course of the simulations. In the second row, the same simulations are performed, but in this case the MGE MI<sub>NT</sub> inhibits transformation in the host cell. The bottom two rows show the outcome of simulations in which both MI and MI<sub>NT</sub> infect the same population. (B) Heatmap summarising the outcomes of simulations comparing cell growth patterns with different MI activation patterns: *f*, the normal rate of activation, is either low (0.005) or high (0.5), as is *f*<sub>C</sub>, the rate of activation in C state. (C) Heatmap summarising the same simulations shown in panel B, but for MI<sub>NT</sub>. (D) Competition between MI<sub>NT</sub> operating with its optimal strategy (*f* and *f*<sub>C</sub> low) and MI operating with its optimal strategy (*f* low, *f*<sub>C</sub> high). Both MGEs were allowed to infect the same population in these simulations. The heatmap shows the ratio of strains infected with MI<sub>NT</sub> to those infected with MI when cells grew and expressed competence for transformation under different strategies. Raw data are tabulated in <u>S1 Data</u>.

doi:10.1371/journal.pbio.1002394.g010

advantage to MGEs over competitors in the same cell, regardless of whether they inhibit transformation or not. This contrasts with the disruption of host cell competence, which is an effective strategy for individual MGEs, but provides the same benefit to all MGEs parasitizing the same host. Hence the effectiveness of transformation against MGEs is likely to be partly maintained by competition between MGEs that infect the same cell.

#### Transformable Streptococci Harbour Fewer Prophages

Our hypothesis predicts that parasitic MGEs integrated into chromosomes that have not achieved fixation should be vertically transmitted at a lower rate in recombining populations of naturally transformable bacteria relative to equivalent nontransformable bacteria. The implication is, all else being equal, that transformable bacteria should harbour fewer MGEs. This is difficult to test with draft assemblies of bacterial genomes, which are fragmentary, often particularly so in regions encoding MGEs; this situation is exacerbated when multiple similar elements are present in the same genome [57]. An alternative approach is to use complete and high quality draft genomes from species or genera containing a mix of transformable and nontransformable bacteria. There are over 140 suitable genome assemblies available for streptococci, of which a subset have been demonstrated to be naturally transformable [28,32,84,121-123]; additionally, Streptococcus pseudopneumoniae was assumed to be naturally transformable, based on its close relationship with S. pneumoniae and Streptococcus mitis. In agreement with a previous comment by Beres et al. [124] based on the small number of genomes available at the time, streptococci demonstrated to be naturally transformable had significantly fewer prophages than nontransformable streptococci (mean number of prophages per transformable genome: 0.74; mean number of prophages per nontransformable genome: 1.85; Wilcoxon rank sum test: W = 3065.5, p = 0.00021; <u>S4 Table</u> and <u>S14 Fig</u>) despite tending to have larger genomes (mean size of transformable genome = 2,073,450 bp; mean size of nontransformable genome = 2,002,130 bp; Wilcoxon rank sum test: W = 1702, p = 0.013; <u>S4 Table</u>). However, there are several caveats to such an analysis. Firstly, the sampling of genomes is nonrandom, something that this analysis partly seeks to address (see Methods). Secondly, all samples are filtered by selection, and, therefore, low-fitness isolates that accumulate large numbers of MGEs are likely to be lost from the population quickly, thereby limiting the opportunity for sampling. Thirdly, there are other systems that inhibit the transmission of MGEs, thereby making the comparison uncontrolled; it is noteworthy, for instance, that an analogous comparison of MGEs and CRISPR systems in streptococci did not find evidence for these systems providing a protective effect against phage infection [125]. In another example, a within-species analysis of A. actinomycetemcomitans cells, of which a subset were rendered nontransformable by MGE insertions into the *comM* gene, reported noncompetent isolates to have an increased susceptibility to further MGE infection, although this was confounded by the associated loss of some CRISPR functionality [113].

#### Discussion

The simulations and genomic analyses presented here were motivated by the unanswered question of what function underlies the major evolutionary benefit of genetic transformation. The model structure followed from the assumption that most HDT through transformation would be between clonally related cells, as a consequence of the physical structuring of populations, combined with the observations that imports are skewed toward shorter recombinations [60,85], the import of deletions is highly efficient in the absence of recognition by mismatch repair [59] or restriction-modification systems [89], and prophages are highly variable over short evolutionary timescales [57,101]. The resulting model shows a benefit, at the individual and group levels of selection, when transformation is sufficiently fast and asymmetric to remove deleterious MGEs from chromosomes. This benefit was found to be independent of whether competence was transiently or constitutively expressed; one difference was that intermittent periods of competence alleviated the elimination of some MGEs with beneficial cargo genes. Coordination of competence with cell–cell killing resulted in oscillatory patterns of growth that synergistically inhibited both the vertical and horizontal spread of MGEs. "Bet hedging" could also be effective at inhibiting the transmission of MGEs; elimination of parasitic elements from the chromosome would be beneficial to the competent subpopulation, while the noncompetent, more quickly replicating cells would have the advantage when such MGEs with spores and "persister" cells [42] are better understood.

In order to be plausible, the model depended on clonally related infected and uninfected cells coexisting within populations that could exchange DNA through transformation. This was demonstrated to occur using genomic data on within-host bacterial diversity, albeit limited to a single colony per timepoint. Additionally, a likely example of a detectable transformation event removing a chromosomally integrated MGE from an infected cell was identified (S6 Fig). Even if the rate at which transformation events occur at a given locus per unit time is low, there is the opportunity for a homologous recombination event to be beneficial through removing an inserted MGE up until the point at which it activates or becomes fixed in the local population; this contrasts with the hypothesised role of transformation in the repair of dsDNA breaks, for instance, as in these circumstances homologous recombination is only beneficial if it affects the damaged locus before repair by other means or the next round of chromosomal replication occurs.

However, that our hypothesised mechanism is possible and potentially of benefit to cells does not necessarily imply that it is the primary biological role of transformation. The extent to which cells benefit from this function may be inferred from the countermeasures employed by MGEs to avoid elimination by homologous recombination. It is highly unlikely that any function other than transformation is targeted, based on the plethora of competence genes disrupted by MGE insertions: preventing transcriptional activation of the competence machinery (*comK*); eliminating the major structural component of the pseudopilus necessary for DNA binding (*comYC*), and inhibiting post-binding processing of the DNA (*comFA* and *comM*). The loci at which these MGEs integrate are unlikely to be random [126], particularly in cases such as these in which they are conserved across species and genera despite divergence of the insertion sites (S10 Fig). Additional evidence is the convergent evolution of MGEs inserting into two distinct sites of the *comFA* gene in different genera, driven by divergent integrases with only 34.1% amino acid identity with one another. This is despite MGE insertion sites generally being enriched outside of CDSs [57,126].

Targeting of competence genes for disruption is not the only mechanism by which MGEs inhibit transformation events that may eliminate them from the chromosome. Some encode secreted DNases that degrade exogenous dsDNA. Secreted "streptodornases" have been identified on prophages in a number of streptococcal and lactococcal species [127]. In *B. subtilis*, the prophage-encoded DNase YokF was found to decrease transformation rates by an order of magnitude [128]. A search of *Campylobacter jejuni* isolates that transformed at a reduced rate identified the DNase Dns, encoded by the CJIE1 MGE [129]; the presence of this protein reduced transformation rates by three orders of magnitude. Similarly, two further orthologous DNases were identified in the *C. jejuni* MGEs CJIE2 and CJIE4, which were each capable of reducing transformation frequencies of their host cells by two orders of magnitude [130]. Recently, an ICE in *V. cholerae* encoding the IdeA DNase was found to inhibit transformation by two to three orders of magnitude [131].

These countermeasures are only advantageous to MGEs if transformation inhibits their transmission in natural populations (Fig 7). If the condition that the cost of the competence machinery to cells is less than that of parasitic MGEs is fulfilled (Fig 3C), this implies transformation is likely to provide a net benefit to cells. That MGEs cause a severe cost to their hosts seems a reasonable inference, given the presence of defences against horizontal transmission of MGEs. Analogously, these defences are also part of an arms race, given the existence of MGEencoded proteins that inhibit restriction modification [132] and CRISPR systems [133], alongside the prophage insertion into a CRISPR locus identified in this work. Therefore, while MGE removal may not be the only activity facilitated by the competence system, there is evidence that it is a function that provides a fitness advantage to the cell. Hence, this role alone may be of sufficient selective benefit to drive the evolution of the necessary machinery. However, our model does not predict that MGEs are likely to become sufficiently adept at preventing transformation that they would render it ineffective over long evolutionary timescales. Only a subset of MGEs would be expected to disrupt the activity of the competence machinery, because in so doing they benefit any superinfecting MGEs that would also likely progressively reduce the fitness of their host cell. The accumulation of further deleterious MGEs could potentially be inhibited by a compensatory improvement in defences against horizontal MGE transmission, which could account for the occasional success of nontransformable, clonally evolving lineages such as clade B of BC4-6B, S. pneumoniae CC180 [134], and PMEN2 [84].

The atypical stability of the prophage, and, indeed, the rest of the accessory genome, in these clonal S. pneumoniae lineages is consistent with transformation being important in inhibiting the vertical transmission of viral sequences in bacteria. Nevertheless, there are also alternative mechanisms that can cause the loss of MGEs that should be considered. The first is spontaneous deletion of sequence; this would usually be disadvantageous if occurring at random, as it would remove beneficial sequences far more frequently than detrimental sequences. The second is removal of MGEs by intragenomic recombination events, potentially mediated by the tandem *att* site duplications flanking many mobile elements. However, the length of the *att* sites is generally determined by the MGE, and any *att* sites long enough to trigger these rearrangements would be selected against through their inhibition of vertical MGE transmission. Thirdly, MGEs may drive their own excision. Some conjugative episomes can switch between integrated and extrachromosomal forms [135], the latter of which are resistant to elimination by transformation. However, in the case of prophages, it seems very likely that the majority of excision events result in host cell lysis. This may be inferred from the large population falls observed on mitomycin C addition to lysogenic populations of relevant species [136], and, furthermore, the existence of apparently altruistic cell death from "abortive infection" systems upon phage replication would be undermined if cells frequently survived phage infection [137]. Hence, alongside transformation's efficiency in deleting DNA [91–94], it is also a very effective mechanism of eliminating integrated MGEs; the imported sequence will restore the uninfected insertion site without affecting flanking regions, which are likely to contain beneficial genes, with no dependency on MGE-encoded loci to facilitate the process.

The speed and frequency with which phage infection is observed to occur in *S. pneumo-niae*, with competence genes disrupted by a subset of such events, contrasts with the observed population dynamics following the introduction of the anti-pneumococcal polysaccharide conjugate vaccines [138]. This represents the type of environmental change to which transformation has been predicted to speed adaptation [42,46], as strains targeted by the vaccine can evade the effects of immunisation by means of transformation events that switch the bacterium's serotype through allelic replacement at the relevant genetic locus [139]. There was substantial opportunity for switching to occur, as serotypes targeted by the vaccine nevertheless

persisted for several years after the immunisation programme began [138]. Additionally, there was the appropriate motivation for sequence transfer. Although the fitness disadvantage of serotypes targeted by the vaccine was small enough for them not to be immediately eliminated from the population after immunisation began, these serotypes did eventually disappear, suggesting the benefit of acquiring a nonvaccine serotype was greater than the cost of expressing the competence machinery. However, only a subset of the targeted population showed evidence of diversifying in response to vaccination years after immunization had begun, and the examples of serotype switching that could be thoroughly characterised were found to represent the outgrowth of variants that originated prior to the vaccine's introduction [4,138]. That the non-prophage accessory genomes of pneumococci, including the genes determining serotype, were largely stable post-vaccination [57] confirms it is unlikely that transformation's primary role is in facilitating adaptation through sequence diversification. Furthermore, such a role is not consistent with a gene-centric view of evolution, as if diversification is the primary purpose of transformation, then the fitnesses of all chromosomal genes in a competent cell are reduced as a consequence of them potentially being replaced by a different allele from the pool of exogenous DNA.

By contrast, this "chromosomal curing" model, in which transformation is primarily a mechanism for maintaining the integrity of a cooperating set of self-replicating genes (be they a chromosome, chromid, or plasmid) against invasion by selfish parasites, is consistent with selection at the level of the gene, individual, and group. If the import of DNA from divergent genotypes is sufficiently rare, then the fitness of individual genes is reduced by a negligible degree, as their probability of being replaced with a different allele is low. However, each gene may frequently benefit from the loss of linkage with a genomic parasite. The model is also able to rationalise the counterintuitive cell–cell killing within clonal populations, which should be strongly opposed by kin selection, as a mechanism to mitigate against the external threat of parasitic MGEs.

In this model, exchanges between diverse genotypes can be viewed as an accidental byproduct of otherwise beneficial exchanges between clonally related cells. This does not preclude some such diversification through HDT being advantageous, particularly after filtering by selection. Hence, cocirculating lineages within naturally transformable species may differ in their rates of diversification through transformation by orders of magnitude, without a substantial fitness difference being evident [138,140]. That the most frequent sequence exchanges are between near-isogenic cells also explains how transformable bacteria can import substantial lengths of DNA in minutes, yet maintain pseudoclonal population structures over decades [57]. Rather than this reflecting the rarity of sequence exchange, such population stability may reflect the continual antagonism between different mechanisms of HDT.

#### Methods

#### Description of the Microevolutionary Model of HDT

We developed a stochastic compartmental model that included four types of compartments: cells, MGEs, DNA, and a signalling molecule, "C signal." The overall structure of the model is displayed in <u>S15 Fig</u>.

Bacterial cell growth (green arrows in <u>S15 Fig</u>) followed a logistic growth model. In the absence of MGE infections, cells grew at a constant rate  $\gamma$  (set to 0.2  $t^{-1}$ , unless otherwise specified). Analyses of the model output sensitivity to different values of  $\gamma$  are shown in <u>S1</u> and <u>S3</u> Figs. Cells died at a density-dependent rate (brown arrows in <u>S15 Fig</u>) determined by  $\gamma$  and a carrying capacity,  $\kappa$  (10<sup>6</sup> in all simulations). For the *k* cell compartments in the model, the number of cells (*N*<sub>i</sub>) in compartment *i* at time *t* changed at time *t*+1 through the demographic

processes of birth and death by  $P_i$ , which was distributed as:

$$P_{i,t+1} \sim Bin(\gamma.dt, N_{i,t}) - Bin\left(\frac{\gamma \sum_{j=1}^{j=k} N_{j,t}.dt}{\kappa}, N_{i,t}\right)$$

In all simulations, the starting inoculum of each distinct genotype was 100 cells. All cell compartments immutably belonged to one of two strains, each of which could be independently parameterised. The "plastic" aspect of cells' genotypes was defined by two biallelic loci: the first locus could either be "empty" (allele E1) or have an inserted MGE, M1; analogously, the second could be empty (allele E2) or contain a different inserted MGE, M2. Upon density-dependent cell death, one DNA molecule was released from each locus, the type of which depended on the host cell genotype (S15 Fig; blue arrows).

Any cell with an "M1" or "M2" allele therefore carried an MGE and, consequently, grew at a rate  $\gamma(1-c_{\rm M})$ , where  $c_{\rm M}$  was the reduced growth of the host cell owing to the cost of the inserted MGE. This factor only applied to the growth term of the demographic model; cells carrying MGEs were killed through cell density-dependent death at the same rate as noncarriers. Analogously, cells infected with two MGEs grew at a rate  $\gamma(1-c_{\rm M1})$  ( $1-c_{\rm M2}$ ). For cells in the *i*th compartment carrying MGE  $M_{\rm q}$ , the number of MGEs that activated per timestep interval at time *t*,  $A_{\rm q,i,t}$ , was distributed according to the number of cells  $N_{\rm i}$  at time *t* and activation rate of the MGE type *q* in cell type *i*  $f_{\rm q,i}$ :

$$A_{q,i,t} \sim Bin(f_{q,i}.dt, N_{i,t})$$

The number of MGEs of type  $M_q$  released by activations occurring in cell type *i*,  $R_{q,i}$ , at time *t* (dark blue and purple arrows in <u>S15 Fig</u>) depended on the mean burst size,  $b_q$ :

$$R_{q,i,t} \sim Pois(b_q A_{q,i,t})$$

As  $A_{q,t}$  determines  $M_{q,t}$ , rather than  $M_{q,t+n}$  where n > 0, the activation and packaging of MGEs is effectively instantaneous in this model. This means there is no eclipse period. If included, this would limit the fitness of horizontal transfer by slowing the rate of transmission between cells, but as this study focused on inhibition of vertical transmission, it did not form part of this model. The parameter *a* determined the consequence of MGE activation for the cell; if a = 1, then MGE activation killed the host cell, as is typical for prophages; but if a = 0, then MGE activation did not affect the host cell, as is typical for ICEs. As MGE activation involves excision from the host chromosome and packaging of the MGE DNA, any cells killed through this mechanism did not release DNA molecules corresponding to the activated MGE, but the appropriate allele was released from the other locus (dark blue and purple arrows in S15 Fig).

Horizontal DNA transfer was modelled as a two-step process using one of two association parameters:  $\beta$ , the infectivity associated with a particular MGE, and  $\tau$ , a transformation rate associated with a particular cell type. The noncellular components of the model were removed from the environment at a constant "washout" rate,  $\omega$ , set at 0.6  $t^{-1}$  unless otherwise specified (grey arrows in <u>S15 Fig</u>). Analyses of the model output's sensitivity to different values of  $\omega$  are shown in <u>S1</u> and <u>S3</u> Figs. Hence, the overall rate at which noncellular agents were removed from the simulation was a composite of cellular binding and elimination from the extracellular environment. For the *q*th DNA compartment, molecules were removed at a composite rate  $r_q$ representing "washout" and binding to each of the *k* cell compartments, each containing  $N_i$  cells and undergoing transformation at a cell-determined rate  $\tau_i$ :

$$r_q(DNA) = \omega + \sum_{i=1}^{i=k} \tau_i N_i$$

Similarly, for the *q*th MGE compartment, elements were removed at a composite rate  $r_q$  representing washout and binding to each of the *k* cell compartments, each containing  $N_i$  cells, at the MGE-determined rate  $\beta_q$ :

$$r_q(MGE) = \omega + \sum_{i=1}^{i=k} \beta_q N_i$$

For the *q*th type of noncellular agent in the model, the number of agents of that type  $(N_q)$  decreased by  $d_q$  per timestep as determined by the relevant value of  $r_q$ :

$$d_q \sim Bin(1 - e^{-r_q \cdot dt}, N_q)$$

The noncellular agents were then assigned to cell compartments through a multinomial distribution, which allowed for differences in  $\tau$  between cell types. Within each compartment, the bound DNA molecules and MGEs were then randomly assigned to individual cells. In cases in which a single cell was bound to multiple noncellular agents, a single noncellular agent was randomly selected for interaction at that timestep; MGEs interacted through causing infection (maroon arrows in <u>S15 Fig</u>), while DNA interacted through causing a transformation event (orange arrows in <u>S15 Fig</u>). This structure permitted a single association constant to parameterise interactions between noncellular agents and cells in a manner that could be limited by either partner in the interaction. However, this structure has the disadvantage of artefactual antagonism between MGEs and DNA in the cases where both are bound to a single cell, but only one is selected to interact with the cell. This effect is small unless there are large numbers of noncellular agents interacting with individual cells per timestep. Simulations mirroring those in Figs <u>3B</u> and <u>4C</u> were carried out in which cells preferentially interacted with MGEs if DNA was also bound to the cell, rather than the selection being random; the results are shown in <u>S13E and</u> S13F Fig, demonstrating that any artefactual antagonism between DNA transformation and MGE infection was negligible at the *dt* interval used in all simulations  $(10^{-3})$ .

In cases in which MGEs were introduced at a constant rate, such as <u>S13A–S13D</u> Fig, the number of MGEs of type q entering at each timestep,  $E_q$ , was determined by the entry rate,  $e_q$ , and the MGE burst size:

$$E_a \sim Bin(e_a.dt, b_a)$$

The relative rate at which the "M" and "E" alleles at the two loci were exchanged was also determined by the asymmetry parameter,  $\varphi$ . In the case of symmetrical transformation ( $\varphi = 1$ ), the rate at which the *i*th cellular compartment underwent transformation was determined by  $\tau_i$  and the number of available DNA molecules. These factors also determined the rate at which any transformation in which the donor and recipient alleles were the same occurred; such recombinations did not affect cell genotype, but nevertheless depleted DNA molecules. When  $\varphi > 1$ , favouring the import of longer alleles, if  $B_{i,q}$  complexes of a DNA molecule of the *q*th compartment, corresponding to an "E" allele DNA molecule, were bound to a cell of the *i*th compartment, with an "M" allele at the relevant locus, the number of transformation events  $T_{i,q}$ 

was distributed as:

$$T_{i,q} \sim Bin(\varphi^{-1}, B_{i,q})$$

Correspondingly, when  $\varphi < 1$ , favouring the import of shorter alleles, if  $B_{i,q}$  complexes of a DNA molecule of the *q*th compartment, corresponding to an "M" allele DNA molecule, bound to a cell of the *i*th compartment, with an "E" allele at the relevant locus, the number of transformation events  $T_{i,q}$  was distributed as:

$$T_{i,q} \sim Bin(\varphi, B_{i,q})$$

In all other cases,  $T_{i,q} = B_{i,q}$ . The default value of  $\varphi$  used in these simulations, 0.1, is a conservative estimate assuming a geometric distribution of imported DNA lengths parameterised according to the typical length of shorter classes of MGEs (~15 kb) [57] and an estimate of the transformation length distribution biased away from shorter transformation events (mean length of ~6.6 kb) [4,60].

The transformation rate also varied through the simulations in cases where the "C state" was included (S15 Fig), representing the regulated, transient competence state observed in many bacterial species. The trigger for entering C state was the "C signal," the levels of which (S) were determined by the rate of production by all k compartments of cells at a rate of  $e_{\rm C}$  per cell (set to 10  $t^{-1}$  per cell; light green arrows in S15 Fig), and elimination at the extracellular washout rate,  $\omega$  (grey arrow in S15 Fig):

$$S_{t+1} \sim S_t + Pois(e_C \sum_{i=1}^{i=k} N_{i,t}.dt) - Bin(\omega.dt, S_t)$$

When the C signal surpassed the threshold  $t_{\rm C}$  (set to 10<sup>7</sup>), the number of non-C-state cells (*N*) of compartment *i* entering C state ( $C_i^{\rm entry}$ ) was distributed according to a rate  $g_{\rm C}$  (set to 10  $t^{-1}$  unless stated; red arrow in <u>S15 Fig</u>):

$$C_i^{entry} \sim Bin(g_c.dt, N_i)$$

Independently of the C signal, cells of compartment *i* in C state ( $C_i$ ) exited C state at a rate  $r_C$  (set between 0 and 0.9  $t^{-1}$ , depending on the growth pattern):

$$C_i^{exit} \sim Bin(r_c.dt, C_i)$$

Cells in the C state grew at a reduced rate  $\gamma(1-c_C)$ , in which  $c_C$  was the cost associated with the C state. In oscillatory growth patterns, C-state cells were subject to growth arrest ( $c_C = 1$ ), as has been observed in some species in which competence is regulated [37,38]. C-state cells underwent transformation at a higher rate ( $\tau_C$ ) than cells not in C state (orange arrows in S15 Fig), and drove cell–cell killing of cells not in the "C state" as a mass action process at a rate  $k_C$ (dashed lines in S15 Fig); this mirrors "fratricide" in some streptococcal species or "cannabilism" in *B. subtilis* [15]. For the *k* cell compartments in the model, the number of non-C-state cells ( $N_i$ ) in the *i*th compartment killed by this mechanism ( $K_i$ ) depended on the total population of C-state cells across all compartments ( $C_j$ ), as well as  $k_C$ :

$$K_i \sim Bin(k_C \sum_{j=1}^{j=k} C_j.dt, N_i)$$

Therefore, the overall change in the population of compartment *i*, if such cells were not in the C state, could be summarised as:

$$N_{i,t+1} = N_{i,t} + P_{i,t} - K_{i,t} - A_{i,t} - C_{i,t}^{entry} + C_{i,t}^{exit} - T_{i \to !i,t} + T_{!i \to i,t} - M_{i \to !i,t} + M_{!i \to i,t}$$

where  $P_i$  is the demographic change resulting from cell growth and death;  $K_i$  is the reduction as a consequence of cell-cell killing;  $A_i$  is the loss of cells due to MGE activation;  $C_i^{\text{entry}}$  is the number of cells entering C state, while  $C_i^{\text{exit}}$  is the number of cells of the same genotype exiting C state;  $T_{i \rightarrow i}$  is the consequence of transformation driving divergence to different genotypes, while  $T_{!i \rightarrow i}$  is the number of cells being converted to compartment *i* by transformation; and  $M_{i \rightarrow !i}$  represents MGE infection converting compartment *i* cells to other genotypes, whereas  $M_{!i \rightarrow i}$  is the reciprocal conversion of other genotypes to compartment *i* through infection. If compartment *i* corresponds to C-state cells, the formula for the population change no longer includes cell-cell killing, and the impact of the state change terms is reversed:

$$C_{i,t+1} = C_{i,t} + P_{i,t} - A_{i,t} - C_{i,t}^{exit} + C_{i,t}^{entry} - T_{i \to !i,t} + T_{!i \to i,t} - M_{i \to !i,t} + M_{!i \to i,t}$$

For MGEs of compartment  $q(M_q)$  interacting with k cell compartments, the change between timesteps can be summarised as:

$$M_{q,t+1} = M_{q,t} + R_{q,t} + E_{q,t} - d_q$$

where  $R_q$  represents the release of MGEs from host cells through activation,  $E_q$  is the spontaneous entry of MGEs into the model, and  $d_q$  represents MGEs lost to infection of cells and MGE washout and degradation.

For DNA of compartment q ( $D_q$ ) interacting with k cell compartments, the change between timesteps can be summarised as:

$$D_{q,t+1} = D_{q,t} + L_{q,t} - d_q$$

where  $L_q$  represents the release of DNA through cell lysis (either a consequence of cell-densitydependent death, MGE activation or cell–cell killing), and  $d_q$  represents the uptake of DNA by competent cells and DNA washout and degradation.

The model was implemented using C++, with the GNU scientific library. Each simulation was run from t = 0 to t = 1,000, with  $10^3$  timesteps per unit time. Neither increasing the number of timesteps per unit time 10-fold, nor doubling the endpoint value of t, substantially altered the displayed results of simulations involving different MGEs and growth patterns. In some simulations involving high rates of transformation ( $\tau \ge 0.01$ ), the number of timesteps per unit time resulted in detectable approximation errors in some model compartments, but validatory simulations confirmed this did not affect reported summary statistics. At t = 0, each genotype started at a frequency of 100 cells. If a single MGE featured in the simulation, then the two genotypes were uninfected and infected; if two MGEs featured in the simulation, then the starting cells were the two singly infected genotypes. Within each timestep, all cells were considered capable of replication at the appropriate growth rate. Molecules were first bound to cells and underwent transformation and MGE infection. Those cells remaining unbound to molecules entered and exited the C state, were eliminated through cell-cell killing, activation of some MGEs, or died through cell-density-dependent death at the appropriate per capita rates. Reported summary values were calculated over the full extent of the simulation and represent the mean of three simulations. The source code is available for download from https://github. com/nickjcroucher/mgeTransformation. Parameters are summarised, along with typical values, in <u>S1 Table</u>; simulation outputs are recorded in <u>S1 Data</u>.

#### Characterising Within-Host Pneumococcal Variation

The original dataset comprised 3,085 de novo assemblies of pneumococcal isolates from the Mae La refugee camp [98]. In order to detect short-term changes in mobile genetic element content, this study identified 374 hosts associated with two or more isolates of the same multilocus sequence type [98] and considered all 1,751 genomes from bacteria carried by these individuals. A quality threshold of a draft assembly N50 greater than 10 kb and a total draft assembly length between 1.75 Mb and 2.75 Mb was imposed on this set; genomes that did not meet these criteria were reassembled with Velvet as described previously [4]. This produced a final set of 1,715 sequences from 371 hosts. CDSs within these genomes were annotated using Prodigal [141] with a model trained on the reference sequence of S. pneumoniae ATCC 700669 [142] and translated to generate a database of 3,660,212 proteins. Using BLASTP [143] with an E-value threshold of  $10^{-10}$ , this database was searched with a single representative sequence from each of the 355 clusters of orthologous proteins previously found to be specific for ICEs [57]. This process was repeated using 590 proteins specific for prophages, 142 proteins specific for phage-related chromosomal islands, and three proteins specific for a particular prophage remnant [57]. Putative variation in MGE content was inferred where two isolates of the same sequence type varied by at least five BLASTP matches to proteins characteristic of a single type of MGE; this identified 281 hosts with candidate short-term accessory genome variation. The original Illumina sequence reads were then mapped against the variable protein coding sequences using BWA [144]. This allowed the many cases likely representing the variable results of de novo assembly to be distinguished from genuine cases of MGE acquisition or loss. Of the genuine cases, almost all corresponded to changes in prophage content.

In order to determine whether such changes were likely to have occurred within a single carriage episode, it was necessary to construct a phylogeny to determine the level of relatedness between isolates from the same host. This required focusing on lineages commonly identified within the dataset; based on the candidate instances of within-host MGE variation, BCs 1-19F and 4-6B were selected (S3 Table), along with one particular case in BC14. A reference genome assembly was needed for each BC being analysed; S. pneumoniae Taiwan<sup>19F</sup>-14 was appropriate for BC1-19F [145], whereas novel references were required for the other two clusters. These were constructed by combining the original Velvet assemblies [98] with SGA assemblies [146] using Zorro [147], then ordering the contigs using ACT [148], as described previously [138]. For all cases in which at least two isolates of the relevant BAPS cluster had been isolated from a single host, the Illumina reads were then mapped against the reference sequence using SMALT  $[\underline{149}]$  as described previously  $[\underline{138}]$ . The resulting whole genome alignment was then analysed using Gubbins [99] to generate a maximum likelihood phylogeny while accounting for the frequent transformation events occurring in pneumococcal lineages. These analyses each corresponded well with the isolates' metadata, identifying closely related clusters of isolates from individual hosts that represented likely individual carriage episodes.

In the instances in which these matched probable cases of MGE variation, regions of similarity between the de novo assemblies of the relevant isolates were identified through a BLAT [150] comparison of all the contigs in each sequence, using standard settings. This comparison file was then used to inspect the assemblies using ACT [148]. The originally identified MGE-associated sequences within the assemblies were then located, and, if part of a larger insertion that had characteristics of an MGE, the element was manually annotated. In some cases, genomes had to be reassembled as described for the references, then organised into scaffolds with SSPACE2 [151] as described previously [138], in order to extract the relevant MGE sequence. This also allowed their insertion sites to be ascertained and classified as described previously [57]. These annotated MGEs have been submitted to Genbank with the accession codes listed in <u>S2 Table</u>. To ensure they represented genuine pneumococcal prophages, they were included in a hierarchical clustering of known pneumococcal prophages, constructed based on CDS content as described previously [134]. Sequence reads were then mapped against these prophages using BWA with a requirement for absolute sequence identity for alignment, to minimise mapping of sequence reads originating from other prophages not in the reference set. These read alignments were then used to generate the heatmaps shown in Figs <u>6</u>, <u>8</u>, <u>S5</u>, <u>S6</u> and <u>S8</u>.

# Distribution of Transformation and MGEs between Streptococcal Species

The 143 complete or high-quality draft streptococcal genomes listed in <u>S4 Table</u> were scanned for prophages using Phage\_Finder v2.1 [152]. Naturally transformable species within the streptococcal genus were identified based on past reviews and recent experimental work [28,32,84,121–123]. The comparison of these raw data found naturally transformable isolates to have significantly fewer prophages than nontransformable isolates (mean number of prophages per transformable genome: 0.74; mean number of prophages per nontransformable genome: 1.77; Wilcoxon rank sum test: W = 3316, p = 0.00089). To ensure this result was not a consequence of biased sampling, three genomes from unnamed species were excluded (*Streptococcus* sp. I and *Streptococcus* sp. VT), as their competence for transformation could not be established. Pneumococcal isolates that represented duplicate samples of very closely related genotypes were also removed (*S. pneumoniae* R6, a duplicate of D39 [153], and *S. pneumoniae* 03–4156, 03–4183, 99–4038, and 99–4039, which all share a prophage insertion with the closely related isolate OXC141 [134]). The results described in the Discussion confirm that the observed association persists in this curated dataset.

#### **Supporting Information**

S1 Data. Summarised values output from simulations used to generate heatmaps in Figs 1, 3, 4, 5, 7, 10, S1-S3 and S13. (XLSX)

S1 Fig. The effects of changing noncellular component washout rates and cell growth rates on the spread of MGEs between constitutively competent cells. (A) This heatmap is displayed as in Fig 3B, but with the rate at which noncellular components are washed out reduced by an order of magnitude to  $\omega = 0.06$ . (B) This heatmap is displayed as in Fig 3B, but with the rate at which noncellular components are washed out increased to  $\omega = 0.99$ . (C) This heatmap is displayed as in Fig 3B, but with the cell growth rate  $\gamma$  halved to 0.1. (D) This heatmap is displayed as in Fig 3B, but with the cell growth rate  $\gamma$  doubled to 0.4. Raw data are tabulated in S1 Data.

(PDF)

#### S2 Fig. Effects of the frequency and amplitude of cell population oscillations on MGE

**transmission.** (A) This heatmap summarises simulations in which the amplitude ( $k_c$ ) and frequency ( $r_c$ ) of cell population oscillations was varied. The colour of the cells represents the proportion of the cell population infected with MGEs over the course of the simulations. Each cell is split into two components based on the speed with which the strain entered the C state ( $g_c = 1 \text{ or } 10$ ). In this panel, the MGE present was MV ( $\beta = 10^{-3}$ ), and transformation was parameterised as  $\tau = 10^{-4}$  and  $\varphi = 0.5$ . (B) This heatmap is displayed as in panel A, but the MGE present was MV ( $\beta = 10^{-1}$ ) and transformation was parameterised as  $\tau = 10^{-3}$  and  $\varphi = 0.5$ . (C) This heatmap is displayed as in panel A, but the MGE present was MH ( $\beta = 5x10^{-7}$ ) and transformation was parameterised as  $\tau = 10^{-6}$  and  $\varphi = 10^{-1}$ . (D) This heatmap is displayed as in panel A,

but the MGE present was MH ( $\beta = 10^{-6}$ ) and transformation was parameterised as  $\tau = 10^{-3}$  and  $\varphi = 10^{-1}$ . Raw data are tabulated in <u>S1 Data</u>. (PDF)

**S3 Fig. HDT between transiently competent cells.** Panels A–D show the effects of changing noncellular component washout rates and cell growth rates on the transmission of MGEs between transiently competent cells. (A) This heatmap is displayed as in Fig 4C, but with the rate at which DNA molecules and MGEs were washed out reduced by an order of magnitude to  $\omega = 0.06$ ; the C signal was still washed out at  $\omega = 0.6$  to avoid changing the pattern of bacterial growth. (B) This heatmap is displayed as that in Fig 4C, but with the rate at which DNA molecules and MGEs were washed out increased to  $\omega = 0.99$ ; again, the C signal was still washed out at  $\omega = 0.6$  to avoid changing the pattern of bacterial growth. (C) This heatmap is displayed as in Fig 4C, but with the cell growth rate  $\gamma$  halved to 0.1. (D) This heatmap is displayed as in Fig 4C, but with the cell growth rate  $\gamma$  doubled to 0.4. Panels E and F show the effect of oscillatory growth on the competition between two strains entering and leaving C state in synchrony, but with only one of the strains undergoing transformation in the C state ( $g_{\rm C}$  = 10 and  $r_{\rm C} = 0.5$  in both cases). HDT occurs both symmetrically ("S" columns) and asymmetrically ("A" columns). (E) This heatmap is displayed as in Fig 1A. It shows the outcome of simulated competition between two strains, only one of which is competent for transformation in the C state, undergoing small population oscillations owing to a C-state-associated cell-cell killing rate of  $k_{\rm C} = 10^{-6}$ . (F) This heatmap is displayed as that in <u>Fig 1A</u>. It shows the outcome of simulated competition between two strains, only one of which is competent for transformation in the C state, undergoing large population oscillations owing to a C-state-associated cellcell killing rate of  $k_{\rm C} = 10^{-3}$ . Raw data are tabulated in <u>S1 Data</u>. (PDF)

S4 Fig. Phylogenetic analysis of BC1-19F isolates from longitudinally sampled hosts using Gubbins. (A) Maximum likelihood phylogeny of isolates based on point mutations outside of putative recombination events. Each leaf node is labelled to indicate whether the *comYC* gene, required for efficient transformation, is intact. (B) Annotation of the reference genome of *S. pneumoniae* Taiwan<sup>19F</sup>-14. Mobile genetic element-related sequences (the Tn916-type ICE, PRCIs, and Pneumococcal Pathogenicity Island 1, PPI-1) are marked, as are loci encoding major antigens (the capsule polysaccharide synthesis, *cps*, locus, as well as *pspA* and *pspC*). (C) Putative recombinations occurring during the evolutionary history of BC1-19F. Red blocks represent putative recombinations reconstructed as occurring on an internal branch, which are, therefore, shared by multiple isolates through common descent. Blue blocks represent putative recombinations reconstructed as occurring on a terminal branch, and are, therefore, unique to a single isolate.

(PDF)

**S5 Fig. Distribution of prophage sequences within BC1-19F.** (**A**) Maximum likelihood phylogeny generated by Gubbins, as displayed in <u>S4 Fig</u>. (**B**) Hierarchical clustering of prophages identified within BC1-19F and BC4-6B with previously identified pneumococcal prophages, based on CDS content. Tips with dashed lines represent those prophages identified within BC1-19F. (**C**) CDS annotations of the 14 prophages extracted from representatives of BC1-19F. (**D**) Bars marking the extent of the individual prophage, coloured to represent their site of insertion within the pneumococcal chromosome. Vertical lines within these bars represent breaks between contigs. (**E**) Heatmap representing the distribution of prophage sequences across BC1-19F. Each row corresponds to an isolate in the phylogeny and is coloured blue where there is a low depth of sequence read mapping (indicating the sequence is absent from

the isolate's genome) and red where there is a high depth of sequence read mapping (indicating the sequence is present in the isolate's genome). Due to sequence similarity between prophages, there is extensive crossmapping between related MGEs. Each case of *comYC* disruption can be associated with the insertion of a prophage into the gene. (PDF)

S6 Fig. Apparent removal of an MGE through an interstrain transformation event. (A) Maximum likelihood phylogeny of BC14 representatives isolated from longitudinally sampled hosts based on point mutations outside of putative recombination events. Each leaf node is labelled to indicate whether the comYC gene is intact. Seven transformable closely related isolates from host ARI-0248 are annotated. (B) Distribution of the putative PRCI PRCI<sub>ARI-0248</sub> between the seven isolates from host ARI-0248, arranged by date of isolation. Each row beneath the PRCI annotation is a heatmap showing the depth of read coverage across the MGE sequence. This indicates the PRCI is absent from two isolates, 09B10533 and 09B13198. (D) Alignment of a putative PRCI from S. pneumoniae TIGR4 with the draft reference genome of S. pneumoniae 10B00189, which carries PRCIARI-0248, and is, in turn, aligned with the draft genome of S. pneumoniae 09B13198, which does not. In both draft genomes, the alternating orange and brown boxes indicate different contigs within the assemblies. Red bands link regions of sequence similarity, as calculated using BLAT; the intensity of the colour represents the extent of the similarity. The green box demarcates the extent of an interstrain transformation event, relative to the reference genome of 10B00198, shared by 09B10533 and 09B13198 (and no other isolates) based on the Gubbins analysis. The recombination spanned PRCI<sub>ARI</sub> <sub>0248</sub> and appears to have caused its deletion in these two isolates. (PDF)

**S7 Fig. Phylogenetic analysis of BC4-6B isolates from longitudinally sampled hosts using Gubbins.** (A) Maximum likelihood phylogeny of isolates based on point mutations outside of putative recombination events. Each leaf node is labelled to indicate whether the *comYC* gene, required for efficient transformation, is intact. (B) Annotation of the reference genome of *S. pneumoniae* 10B02680. Alternating orange and brown blocks represent different ordered contigs in the curated de novo draft assembly. Mobile genetic element-related sequence (the ICE, PRCIs, prophages, and PPI-1) are marked, as are loci encoding major antigens (the capsule polysaccharide synthesis, *cps*, locus, as well as *pspA* and *pspC*). (C) Putative recombinations occurring during the evolutionary history of BC4-6B. Red blocks represent putative recombinations reconstructed as occurring on an internal branch, which are, therefore, shared by multiple isolates through common descent. Blue blocks represent putative recombinations reconstructed as occurring on a terminal branch and are, therefore, unique to a single isolate. (PDF)

**S8 Fig. Distribution of prophage sequences within BC4-6B.** (A) Maximum likelihood phylogeny generated by Gubbins, as displayed in <u>S7 Fig</u>. (B) Hierarchical clustering of prophages identified within BC1-19F and BC4-6B with previously identified pneumococcal prophages, based on CDS content. Tips with dashed lines represent those prophages identified within BC4-6B. (C) CDS annotations of the twelve prophages extracted from representatives of BC4-6B. (D) Bars marking the extent of the prophages, coloured to represent their site of insertion within the pneumococcal chromosome. Vertical lines within these bars represent breaks between contigs. (E) Heatmap representing the distribution of prophage sequences across BC4-6B. Each row corresponds to an isolate in the phylogeny and is coloured blue where there is a low depth of sequence read mapping and red where there is a kigh depth of sequence read mapping. Due to sequence similarity between prophages, there is extensive crossmapping

between related MGEs. Each case of *comYC* disruption can be associated with the insertion of a prophage into the gene.

(PDF)

S9 Fig. Prophages with integrases similar to that found in the prophage disrupting comYC in S. pneumoniae 670-6B (SP670 2190). (A) Comparison of Streptococcus mutans isolates UA159 and NLML9, the latter of which has a prophage inserted into the *comYC* gene encoding the major structural component of the competence pilus. The accession codes of each sequence are given in brackets underneath the isolate names. Blue and orange boxes represent cellular CDSs, with the direction of transcription indicated by their vertical position relative to the horizontal line; pink boxes represent MGE CDSs in the same way. Brown boxes linked by dashed lines mark fragments of a pseudogene disrupted by an MGE insertion. The red bands link regions of similar sequence in the two loci, with the intensity of the colour representing the strength of the match. The level of protein identity between this prophage integrase and that disrupting the comYC gene of S. pneumoniae 670-6B (SP670\_2190) is annotated. (B) Comparison of Streptococcus parauberis isolates KRS-02109 and KRS-02083, the latter of which has a prophage inserted into the *comYC* gene. (C) Comparison between Lactococcus lactis isolates IL1403 and KLDS 4.0325, the latter of which has a prophage inserted into the *comYC* gene. This comparison is also shown in Fig 9A. (D) Comparison between Streptococcus agalactiae isolates COH1 and FSL S3-277, the latter of which has a prophage inserted into the cas3 gene of the S. agalactiae CRISPR2 locus. This comparison is also shown in Fig 9B.

(PDF)

**S10 Fig. MGE insertion sites within competence-associated genes.** (A) Insertion of prophages into *comYC*. All prophages had an integrase similar to SP670\_2190. This section of the *comYC* codon alignment shows the prophages identified in *Streptococcus parauberis*, *Streptococcus mutans*, and *Lactococcus lactis* all insert into an orthologous, but not perfectly conserved, location within the gene. (B) Insertion of MGEs into *comM*. All MGEs had an integrase similar to CF65\_00446. This section of the *comM* codon alignment shows the MGEs identified in *Pseudomonas syringae*, *Francisella philomiragia*, *Mannheimia haemolytica*, and *Acinetobacter baumannii* all insert into an orthologous, but not perfectly conserved, location within the gene. (C) Insertion of prophages into *comFA*. The prophages identified in *Bacillus thuringiensis* and *Bacillus cereus* have integrases similar to LMRG\_01511 (and are 80.9% identical to one another), and both insert at orthologous, but nonidentical, sites within the *comFA* codon alignment. However, the prophage inserted into *comFA* in *Streptococcus suis* has a distinct integrase (only 34.1% identity with that identified in *B. cereus*), and correspondingly inserts into a different site much further downstream in the codon alignment.

**S11 Fig. Prophages with integrases similar to that found in the prophage disrupting** *comK* **in** *Listeria monocytogenes* **10403S (LMRG\_01511). (A)** Comparison of *Listeria innocua* isolates 9KSM and Clip11262, the latter of which has a prophage inserted into the *comK* gene, encoding the orthologue of the main regulator of competence in *Bacillus subtilis*. The comparison is displayed as described in <u>S9 Fig. (B)</u> Comparison of *Bacillus cereus* isolates MHI 226 and VD214, the latter of which has a prophage inserted into the *comFA* gene at a site distinct from that targeted by the prophage displayed in <u>Fig 9C</u>. This comparison is also shown in <u>Fig 9D</u>. (C) Comparison of *Bacillus thuringiensis* isolate BMB171 and a representative of serovar tolworthi, the latter of which has a prophage inserted into the *comFA* gene. (D) Comparison of *Enterococcus faecalis* isolates V583 and RMC65, the latter of which

has a prophage inserted into the *radC* gene, often upregulated during competence in multiple species.

(PDF)

**S12** Fig. MGEs with integrases similar to that found in the MGE disrupting *comM* in *Aggregatibacter actinomycetemcomitans* HK1651 (CF65\_00446). (A) Comparison of *Acinetobacter baumannii* isolates LAC-4 and 1598530, the latter of which has an MGE inserted into a CDS encoding an orthologue of ComM, a protein identified as increasing transformation efficiency in *H. influenzae*. The comparison is displayed as described in S9 Fig. (B) Comparison of *Manheimia haemolytica* isolates D171 and USMARC-185, the latter of which has an MGE inserted into a CDS encoding an orthologue of ComM. (C) Comparison of *Francisella philomiragia* isolates ATCC 25015 and FAJ, the latter of which has an MGE inserted into a CDS encoding an orthologue of *Pseudomonas syringae* isolates UMAF0158 and BRIP34881, the latter of which has an MGE inserted into a CDS encoding an orthologue of ComM.



S13 Fig. Exploring parameter variation and model limitations relating to interactions between MGEs and cells. Panels A-D show further simulations investigating MGE strategies for reducing elimination by transformation. A particular issue with the simulations presented in Fig 10B and 10C was that the high value of f was especially detrimental to an MGE in early timesteps, when cells are at a low density; these simulations test for the success of different strategies when MGEs are able to invade a cell population after it had reached its carrying capacity. (A) Heatmap showing the same simulations as in Fig 10A, except that bursts of MGEs were introduced at a rate of  $10^{-3} t^{-1}$  rather than being polymorphic in the initial population. The colours of the cells represent the proportion of the cell population infected by MGEs over the duration of the simulations. (B) Heatmap showing the overall cell population through the simulations shown in panel A on a  $\log_{10}$  scale. (C) Heatmap showing the same simulations as in Fig 10B, except that bursts of MGEs were introduced at a rate of  $10^{-3} t^{-1}$  rather than being polymorphic in the initial population. (D) Heatmap showing the overall cell population through the simulations shown in panel C on a  $\log_{10}$  scale. Panels E and F evaluate the impact of artefactual antagonism between MGE infection and transformation. The model was altered such that whenever cells bound both DNA and MGEs, MGE infection occurred preferentially in place of transformation. (E) The set of simulations displayed in Fig 3B are repeated with the altered model. (F) The set of simulations displayed in Fig 4C are repeated with the altered model. Raw data are tabulated in S1 Data. (PDF)

#### S14 Fig. Distribution of prophages in complete or high-quality draft streptococcal

**genomes.** The genomes listed in <u>S4 Table</u> are plotted in terms of their overall size and the number of prophages detected within them. Points are coloured red if the isolate was known to be naturally transformable, or otherwise blue. (PDF)

**S15 Fig. Structure of the stochastic compartmental model.** (A) Links between cellular, DNA, and MGE compartments in the basic model. Each compartment type is represented by a different colour; the cell genotype can change through either interaction with a DNA compartment (transformation) or an MGE compartment (MGE infection). Not shown are genetically "silent" transformation and infection events that deplete noncellular compartment populations but do not affect cell genotypes. Cells replicate according to their growth rate, as modified by the cost of carried MGEs, and die through density-dependent cell death and activation of some MGEs.

Density-dependent cell death releases one DNA molecule of the allele present at each locus of the genotype; cell deaths associated with MGE activation release a burst of MGEs, and one molecule of the allele present at the nonactivating locus. (**B**) Incorporation of transient competence into the model. All cells generate a C signal, and above a threshold level, this signal drives cells to enter the C state. Cells left C state at a constant per capita rate, independent of the level of C signal. Genetic alterations through transformation were only possible when cells were in the C state. The C state also affected the population dynamics, as, in some simulations, the replication of cells was transiently arrested while they were in C state (if  $c_{\rm C} = 1$  in the "bet hedging" and oscillatory growth patterns), and C-state cells also inhibited the growth of non-C-state cells through cell–cell killing (if  $k_{\rm C} > 0$  in oscillatory growth patterns). (PDF)

## **S1** Table. Description of model parameters with typical values. (DOCX)

**S2 Table.** Properties and accession codes of prophages identified as part of this work. The annotated prophage sequences shown in <u>S5</u> and <u>S8</u> Figs have been deposited in Genbank with the listed accession codes. The insertion sites of the prophages, described as in [57], are detailed along with the properties of the host bacterium. (DOCX)

S3 Table. Epidemiological information association with the sequence data displayed in Figs  $\underline{6}$  and  $\underline{8}$ .

(XLSX)

**S4 Table. Distribution of prophages in streptococcal genomes.** This table displays the properties of annotated streptococcal genomes, whether the isolate is known to be naturally transformable, and the summarized output of the Phage\_Finder algorithm when applied to this sequence. These data were used to test for any difference in the distribution of prophages between isolates known to be naturally transformable and those that are not. (DOCX)

#### Acknowledgments

We are grateful for the support of the Sanger Institute core sequencing and informatics teams, and for the opportunity to discuss this work at the PERMAFROST workshop.

#### **Author Contributions**

Conceived and designed the experiments: NJC SDB PT CF. Performed the experiments: NJC CF. Analyzed the data: NJC RM CW CF. Contributed reagents/materials/analysis tools: PT SDB. Wrote the paper: NJC CF.

#### References

- Fraser C, Hanage WP, Spratt BG. Recombination and the nature of bacterial speciation. Science. 2007; 315: 476–480. 315/5811/476 PMID: <u>17255503</u>
- 2. Ochman H, Lawrence JG, Groisman EA. Lateral gene transfer and the nature of bacterial innovation. Nature. 2000; 405: 299–304. doi: 10.1038/35012500 PMID: 10830951
- Neu HC. The crisis in antibiotic resistance. Science. 1992; 257: 1064–1073. doi: <u>10.1126/science.</u> <u>257.5073.1064</u> PMID: <u>1509257</u>
- Croucher NJ, Harris SR, Fraser C, Quail MA, Burton J, van der Linden M, et al. Rapid pneumococcal evolution in response to clinical interventions. Science. 2011; 331: 430–434. doi: <u>10.1126/science.</u> <u>1198545</u> PMID: <u>21273480</u>

- 5. McInerney JO, Cotton JA, Pisani D. The prokaryotic tree of life: past, present. . . and future? Trends Ecol Evol. 2008; 23: 276–281. doi: 10.1016/j.tree.2008.01.008 PMID: 18367290
- Baltrus DA. Exploring the costs of horizontal gene transfer. Trends Ecol Evol. 2013; 28: 489–495. doi: 10.1016/j.tree.2013.04.002 PMID: 23706556
- Szathmáry E, Maynard Smith J. From replicators to reproducers: the first major transitions leading to life. J Theor Biol. 1997; 187: 555–571. doi: 10.1006/jtbi.1996.0389 PMID: 9299299
- Guglielmini J, Quintais L, Garcillán-Barcia MP, de la Cruz F, Rocha EPC. The repertoire of ICE in prokaryotes underscores the unity, diversity, and ubiquity of conjugation. PLoS Genet. 2011; 7: e1002222. doi: 10.1371/journal.pgen.1002222 PMID: 21876676
- Canchaya C, Fournous G, Chibani-Chennoufi S, Dillmann ML, Brüssow H. Phage as agents of lateral gene transfer. Curr Opin Microbiol. 2003; 6: 417–424. doi: <u>10.1016/S1369-5274(03)00086-9</u> PMID: <u>12941415</u>
- Wilson GG, Murray NE. Restriction and modification systems. Annu Rev Genet. 1991; 25: 585–627. doi: 10.1146/annurev.genet.25.1.585 PMID: 1812816
- Makarova KS, Haft DH, Barrangou R, Brouns SJJ, Charpentier E, Horvath P, et al. Evolution and classification of the CRISPR-Cas systems. Nat Rev Microbiol. 2011; 9: 467–77. doi: <u>10.1038/nrmicro2577</u> PMID: <u>21552286</u>
- 12. Griffith F. The significance of pneumococcal types. J Hyg. 1928; 27: 113–159. PMID: 20474956
- Avery OT, MacLeod CM, McCarty M. Studies on the chemical nature of the substance inducing transformation of pneumococcal types. J Exp Med. 1944; 79: 137–158. PMID: <u>19871359</u>
- 14. Dubnau D. DNA uptake in bacteria. Annu Rev Microbiol. 1999; 53: 217–244. PMID: 10547691
- Johnston C, Martin B, Fichant G, Polard P, Claverys J-P. Bacterial transformation: distribution, shared mechanisms and divergent control. Nat Rev Microbiol. 2014; 12: 181–96. doi: <u>10.1038/nrmicro3199</u> PMID: <u>24509783</u>
- Campbell EA, Choi SY, Masure HR. A competence regulon in Streptococcus pneumoniae revealed by genomic analysis. Mol Microbiol. 1998; 27: 929–939. PMID: <u>9535083</u>
- Pestova E V, Morrison DA. Isolation and characterization of three Streptococcus pneumoniae transformation-specific loci by use of a *lacZ* reporter insertion vector. J Bacteriol. 1998; 180: 2701–2710. PMID: 9573156
- Chen I, Christie PJ, Dubnau D. The ins and outs of DNA transfer in bacteria. Science. 2005; 310: 1456–1460. 310/5753/1456 PMID: <u>16322448</u>
- Kahn ME, Smith HO. Transformation in Haemophilus: a problem in membrane biology. J Membr Biol.; 1984; 81: 89–103. PMID: <u>6387128</u>
- Lacks S, Neuberger M. Membrane location of a deoxyribonuclease implicated in the genetic transformation of *Diplococcus pneumoniae*. J Bacteriol. 1975; 124: 1321–1329. PMID: <u>366</u>
- Londoño-Vallejo JA, Dubnau D. Mutation of the putative nucleotide binding site of the Bacillus subtilis membrane protein ComFA abolishes the uptake of DNA during transformation. J Bacteriol. 1994; 176: 4642–4645. PMID: 8045895
- 22. Gwinn ML, Ramanathan R, Smith HO, Tomb JF. A new transformation-deficient mutant of *Haemophilus influenzae* Rd with normal DNA uptake. J Bacteriol. 1998; 180: 746–748. PMID: 9457884
- **23.** Morrison DA, Guild WR. Transformation and deoxyribonucleic acid size: extent of degradation on entry varies with size of donor. J Bacteriol. 1972; 112: 1157–1168. PMID: 4404818
- 24. Davidoff-Abelson R, Dubnau D. Fate of transforming DNA after uptake by competent *Bacillus subtilis*: failure of donor DNA to replicate in a recombination-deficient recipient. Proc Natl Acad Sci U S A. 1971; 68: 1070–1074. PMID: <u>4995821</u>
- Berge M, Mortier-Barriere I, Martin B, Claverys JP. Transformation of *Streptococcus pneumoniae* relies on DprA- and RecA-dependent protection of incoming DNA single strands. Mol Microbiol. 2003; 50: 527–536. PMID: 14617176
- Mortier-Barriere I, Velten M, Dupaigne P, Mirouze N, Pietrement O, McGovern S, et al. A key presynaptic role in transformation for a widespread bacterial protein: DprA conveys incoming ssDNA to RecA. Cell. 2007; 130: 824–836. S0092-8674(07)00976-2 PMID: 17803906
- Mortier-Barrière I, de Saizieu A, Claverys J-P, Martin B. Competence-specific induction of recA is required for full recombination proficiency during transformation in *Streptococcus pneumoniae*. Mol Microbiol. 1998; 27: 159–70. doi: 10.1046/j.1365-2958.1998.00668.x PMID: 9466264
- Johnsborg O, Eldholm V, Håvarstein LS. Natural genetic transformation: prevalence, mechanisms and function. Res Microbiol. 2007; 158: 767–778. doi: <u>10.1016/j.resmic.2007.09.004</u> PMID: <u>17997281</u>

- 29. Håvarstein LS. Increasing competence in the genus Streptococcus. Mol Microbiol. 2010; 78: 541– 544. PMID: <u>21038480</u>
- Håvarstein LS, Coomaraswamy G, Morrison DA. An unmodified heptadecapeptide pheromone induces competence for genetic transformation in *Streptococcus pneumoniae*. Proc Natl Acad Sci U S A. 1995; 92: 11140–11144. PMID: <u>7479953</u>
- Johnston C, Campo N, Bergé MJ, Polard P, Claverys JP. Streptococcus pneumoniae, le transformiste. Trends in Microbiology. 2014; 22(3):113–119. doi: <u>10.1016/j.tim.2014.01.002</u> PMID: <u>24508048</u>
- Fontaine L, Boutry C, De Frahan MH, Delplace B, Fremaux C, Horvath P, et al. A novel pheromone quorum-sensing system controls the development of natural competence in *Streptococcus thermophilus* and *Streptococcus salivarius*. J Bacteriol. 2010; 192: 1444–1454. doi: <u>10.1128/JB.01251-09</u> PMID: <u>20023010</u>
- **33.** Higgins DA, Pomianek ME, Kraml CM, Taylor RK, Semmelhack MF, Bassler BL. The major *Vibrio cholerae* autoinducer and its role in virulence factor production. Nature. 2007; 450: 883–6. doi: <u>10.</u> <u>1038/nature06284</u> PMID: <u>18004304</u>
- Luo P, Morrison DA. Transient association of an alternative sigma factor, ComX, with RNA polymerase during the period of competence for genetic transformation in *Streptococcus pneumoniae*. J Bacteriol. 2003; 185: 349–58. PMID: <u>12486073</u>
- **35.** Turgay K, Hahn J, Burghoorn J, Dubnau D. Competence in *Bacillus subtilis* is controlled by regulated proteolysis of a transcription factor. EMBO J. 1998; 17: 6730–6738. doi: <u>10.1093/emboj/17.22.6730</u> PMID: <u>9890793</u>
- Claverys J-P, Håvarstein LS. Cannibalism and fratricide: mechanisms and raisons d'etre. Nat Rev Microbiol. 2007; 5: 219–229. PMID: <u>17277796</u>
- Haijema BJ, Hahn J, Haynes J, Dubnau D. A ComGA-dependent checkpoint limits growth during the escape from competence. Mol Microbiol. 2001; 40: 52–64. doi: <u>10.1046/j.1365-2958.2001.02363.x</u> PMID: <u>11298275</u>
- Oggioni MR, Iannelli F, Ricci S, Chiavolini D, Parigi R, Trappetti C, et al. Antibacterial activity of a competence-stimulating peptide in experimental sepsis caused by *Streptococcus pneumoniae*. Antimicrob Agents Chemother. 2004; 48: 4725–4732. doi: <u>10.1128/AAC.48.12.4725-4732.2004</u> PMID: <u>15561850</u>
- Süel GM, Garcia-Ojalvo J, Liberman LM, Elowitz MB. An excitable gene regulatory circuit induces transient cellular differentiation. Nature. 2006; 440: 545–550. doi: <u>10.1038/nature04588</u> PMID: <u>16554821</u>
- Maamar H, Raj A, Dubnau D. Noise in gene expression determines cell fate in *Bacillus subtilis*. Science. 2007; 317: 526–529. doi: <u>10.1126/science.1140818</u> PMID: <u>17569828</u>
- **41.** Johnsen PJ, Dubnau D, Levin BR. Episodic selection and the maintenance of competence and natural transformation in *Bacillus subtilis*. Genetics. 2009; 181: 1521–1533. doi: <u>10.1534/genetics.108</u>. 099523 PMID: 19189946
- Wylie CS, Trout AD, Kessler DA, Levine H. Optimal strategy for competence differentiation in bacteria. PLoS Genet. 2010; 6: e1001108. doi: <u>10.1371/journal.pgen.1001108</u> PMID: <u>20838595</u>
- 43. Redfield RJ. Genes for Breakfast: The Have-Your-Cake and-Eat-It-Too of Bacterial Transformation. J Hered. 1993; 84: 400–404. PMID: <u>8409360</u>
- 44. Fisher RA. The Genetical Theory of Natural Selection. Genetics. 1930; 154: 272.
- 45. Muller H. Some genetic aspects of sex. Am Nat. 1932; 66: 118–138.
- 46. Levin BR, Cornejo OE. The population and evolutionary dynamics of homologous gene recombination in bacteria. PLoS Genet.; 2009; 5: e1000601. doi: <u>10.1371/journal.pgen.1000601</u> PMID: <u>19680442</u>
- **47.** Smith JM. Group Selection and Kin Selection. Nature. 1964; 201: 1145–1147. doi: <u>10.1038/</u> 2011145a0
- Redfield RJ. Evolution of bacterial transformation: is sex with dead cells ever better than no sex at all? Genetics. 1988; 119: 213–221. PMID: <u>3396864</u>
- **49.** Otto SP, Feldman MW. Deleterious mutations, variable epistatic interactions, and the evolution of recombination. Theor Popul Biol. 1997; 51: 134–47. doi: <u>10.1006/tpbi.1997.1301</u> PMID: <u>9169238</u>
- **50.** Redfield RJ, Schrag MR, Dean AM. The evolution of bacterial transformation: sex with poor relations. Genetics. Genetics Soc America; 1997; 146: 27–38.
- Khan AI, Dinh DM, Schneider D, Lenski RE, Cooper TF. Negative epistasis between beneficial mutations in an evolving bacterial population. Science. 2011; 332: 1193–1196. doi: <u>10.1126/science.</u> <u>1203801</u> PMID: <u>21636772</u>

- Chou H-H, Chiu H-C, Delaney NF, Segrè D, Marx CJ. Diminishing returns epistasis among beneficial mutations decelerates adaptation. Science. 2011; 332: 1190–2. doi: <u>10.1126/science.1203799</u> PMID: <u>21636771</u>
- Smith JM. Evolution in Sexual and Asexual Populations. The American Naturalist. 1968; 102 (927):469–473. doi: 10.1086/282559
- 54. Van Valen L. A new evolutionary law. Evol Theory. 1973; 1:1–30.
- 55. Caugant DA, Mocca LF, Frasch CE, Froholm LO, Zollinger WD, Selander RK. Genetic structure of *Neisseria meningitidis* populations in relation to serogroup, serotype, and outer membrane protein pattern. J Bacteriol. 1987; 169: 2781–2792. PMID: <u>3108242</u>
- Musser JM, Kroll JS, Granoff DM, Moxon ER, Brodeur BR, Campos J, et al. Global genetic structure and molecular epidemiology of encapsulated *Haemophilus influenzae*. Rev Infect Dis. 1990; 12: 75– 111. PMID: 1967849
- Croucher NJ, Coupland PG, Stevenson AE, Callendrello A, Bentley SD, Hanage WP. Diversification of bacterial genome content through distinct mechanisms over different timescales. Nat Commun.; 2014; 5: 5471. doi: <u>10.1038/ncomms6471</u> PMID: <u>25407023</u>
- Mostowy R, Croucher NJ, Hanage WP, Harris SR, Bentley S, Fraser C. Heterogeneity in the Frequency and Characteristics of Homologous Recombination in Pneumococcal Evolution. PLoS Genet. 2014; 10: e1004300. doi: <u>10.1371/journal.pgen.1004300</u> PMID: <u>24786281</u>
- Claverys JP, Méjean V, Gasc AM, Sicard AM. Mismatch repair in *Streptococcus pneumoniae*: relationship between base mismatches and transformation efficiencies. Proc Natl Acad Sci U S A. 1983; 80: 5956–60. PMID: <u>6310606</u>
- Croucher NJ, Harris SR, Barquist L, Parkhill J, Bentley SD. A high-resolution view of genome-wide pneumococcal transformation. PLoS Pathog. 2012; 8: e1002745. doi: <u>10.1371/journal.ppat.1002745</u> PMID: <u>22719250</u>
- Ginetti F, Perego M, Albertini AM, Galizzi A. Bacillus subtilis mutS mutL operon: identification, nucleotide sequence and mutagenesis. Microbiology. 1996; 142 (Pt 8: 2021–9. doi: <u>10.1099/13500872-142-</u> 8-2021 PMID: 8760914
- Majewski J, Zawadzki P, Pickerill P, Cohan FM, Dowson CG. Barriers to genetic exchange between bacterial species: Streptococcus pneumoniae transformation. J Bacteriol. 2000; 182: 1016–1023. PMID: 10648528
- Bagci H, Stuy J. A hex mutant of Haemophilus influenzae. Mol Gen Genet MGG.; 1979; 175: 175– 179. doi: 10.1007/BF00425533 PMID: 316097
- Ambur OH, Davidsen T, Frye SA, Balasingham S V., Lagesen K, Rognes T, et al. Genome dynamics in major bacterial pathogens. FEMS Microbiol Rev. 2009; 33: 453–470. doi: <u>10.1111/j.1574-6976.</u> <u>2009.00173.x</u> PMID: <u>19396949</u>
- **65.** Bernstein H, Byers GS, Michod RE. Evolution of Sexual Reproduction: Importance of DNA Repair, Complementation, and Variation. Am Nat.; 1981; 117: 537–549.
- Prudhomme M, Attaiech L, Sanchez G, Martin B, Claverys JP. Antibiotic stress induces genetic transformability in the human pathogen *Streptococcus pneumoniae*. Science. 2006; 313: 89–92. 313/ 5783/89 PMID: 16825569
- **67.** Charpentier X, Kay E, Schneider D, Shuman HA. Antibiotics and UV radiation induce competence for natural transformation in *Legionella pneumophila*. J Bacteriol. 2011; 193: 1114–21. doi: <u>10.1128/JB.</u> 01146-10 PMID: 21169481
- **68.** Dorer MS, Fero J, Salama NR. DNA damage triggers genetic exchange in *Helicobacter pylori*. PLoS Pathog. 2010; 6: 1–10. doi: 10.1371/journal.ppat.1001026
- Wojciechowski MF, Hoelzer M a., Michod RE. DNA repair and the evolution of transformation in Bacillus subtilis. II. Role of inducible repair. Genetics. 1989; 121: 411–422. PMID: <u>2497048</u>
- Hoelzer MA, Michod RE. DNA repair and the evolution of transformation in *Bacillus subtilis*. III. Sex with damaged DNA. Genetics. 1991; 128: 215–223. PMID: <u>1906416</u>
- 71. Redfield RJ. Evolution of natural transformation: testing the DNA repair hypothesis in *Bacillus subtilis* and *Haemophilus influenzae*. Genetics. 1993; 133: 755–761. PMID: <u>8462839</u>
- Michod RE, Wojciechowski MF. DNA repair and the evolution of transformation IV. DNA damage increases transformation. J Evol Biol. 1994; 7: 147–175. doi: <u>10.1046/j.1420-9101.1994.7020147.x</u>
- Boutry C, Delplace B, Clippe A, Fontaine L, Hols P. SOS response activation and competence development are antagonistic mechanisms in *Streptococcus thermophilus*. J Bacteriol. 2013; 195: 696– 707. doi: <u>10.1128/JB.01605-12</u> PMID: <u>23204467</u>
- 74. Redfield RJ. Genes for Breakfast: The Have-Your-Cake and-Eat-It-Too of Bacterial Transformation. J Hered. 1993; 84: 400–404. PMID: 8409360

- **75.** Mongold JA. DNA repair and the evolution of transformation in *Haemophilus influenzae*. Genetics. 1992; 132: 893–898. PMID: <u>1334020</u>
- 76. Sweetman WA, Moxon ER, Bayliss CD. Induction of the SOS regulon of *Haemophilus influenzae* does not affect phase variation rates at tetraneucleotide or dinucleotide repeats. Microbiology. 2005; 151: 2751–2763. doi: <u>10.1099/mic.0.27996-0</u> PMID: <u>16079351</u>
- Charpentier X, Polard P, Claverys JP. Induction of competence for genetic transformation by antibiotics: Convergent evolution of stress responses in distant bacterial species lacking SOS? Curr Opin Microbiol. 2012; 15: 570–576. doi: <u>10.1016/j.mib.2012.08.001</u> PMID: <u>22910199</u>
- Munoz-Najar U, Vijayakumar MN. An operon that confers UV resistance by evoking the SOS mutagenic response in streptococcal conjugative transposon Tn5252. J Bacteriol. 1999; 181: 2782–2788. PMID: <u>10217768</u>
- 79. Konstantinidis KT, Tiedje JM. Trends between gene content and genome size in prokaryotic species with larger genomes. Proc Natl Acad Sci U S A. 2004; 101: 3160–5. doi: <u>10.1073/pnas.0308653100</u> PMID: <u>14973198</u>
- Stewart GJ, Carlson CA. The Biology of Natural Transformation. Annu Rev Microbiol.; 1986; 40: 211– 231. doi: 10.1146/annurev.mi.40.100186.001235 PMID: 3535646
- Macfadyen LP, Chen D, Vo HC, Liao D, Sinotte R, Redfield RJ. Competence development by *Haemo-philus influenzae* is regulated by the availability of nucleic acid precursors. Mol Microbiol. 2001; 40: 700–707. doi: 10.1046/j.1365-2958.2001.02419.x PMID: 11359575
- Antonova ES, Bernardy EE, Hammer BK. Natural competence in *Vibrio cholerae* is controlled by a nucleoside scavenging response that requires CytR-dependent anti-activation. Mol Microbiol. 2012; 86: 1215–31. doi: 10.1111/mmi.12054 PMID: 23016895
- Finkel SE, Kolter R. DNA as a Nutrient: Novel Role for Bacterial Competence Gene Homologs. J Bacteriol. 2001; 183: 6288–6293. PMID: <u>11591672</u>
- Croucher NJ, Hanage WP, Harris SR, McGee L, van der Linden M, de Lencastre H, et al. Variable recombination dynamics during the emergence, transmission and "disarming" of a multidrug-resistant pneumococcal clone. BMC Biol.; 2014; 12: 49. doi: 10.1186/1741-7007-12-49 PMID: 24957517
- Mell JC, Lee JY, Firme M, Sinha S, Redfield RJ. Extensive Cotransformation of Natural Variation into Chromosomes of Naturally Competent *Haemophilus influenzae*. G3 Genes Genomes Genet. Genetics Society of America; 2014; 4: 717–731. doi: <u>10.1534/g3.113.009597</u> PMID: <u>24569039</u>
- Feil EJ, Smith JM, Enright MC, Spratt BG. Estimating recombinational parameters in Streptococcus pneumoniae from multilocus sequence typing data. Genetics. 2000; 154: 1439–1450. PMID: 10747043
- Frost LS, Leplae R, Summers AO, Toussaint A. Mobile genetic elements: the agents of open source evolution. Nat Rev Microbiol. 2005; 3: 722–732. PMID: 16138100
- Majewski J, Cohan FM. The effect of mismatch repair and heteroduplex formation on sexual isolation in Bacillus. Genetics. 1998; 148: 13–18. PMID: 9475717
- Johnston C, Martin B, Granadel C, Polard P, Claverys JP. Programmed Protection of Foreign DNA from Restriction Allows Pathogenicity Island Exchange during Pneumococcal Transformation. PLoS Pathog. 2013; 9: e1003178. doi: <u>10.1371/journal.ppat.1003178</u> PMID: <u>23459610</u>
- Adams A. Transformation and transduction of a large deletion mutation in *Bacillus subtilis*. Mol Gen Genet MGG.; 1972; 118: 311–322. doi: <u>10.1007/BF00333566</u> PMID: <u>4632051</u>
- Stuy JH, Walter RB. Addition, deletion, and substitution of long nonhomologous deoxyribonucleic acid segments by genetic transformation of *Haemophilus influenzae*. J Bacteriol. 1981; 148: 565–571. PMID: <u>6975273</u>
- Claverys JP, Lefevre JC, Sicard AM. Transformation of Streptococcus pneumoniae with S. pneumoniae-lambda phage hybrid DNA: induction of deletions. Proc Natl Acad Sci U S A. 1980; 77: 3534– 3538. doi: 10.1073/pnas.77.6.3534 PMID: 6251465
- Lefevre JC, Mostachfi P, Gasc AM, Guillot E, Pasta F, Sicard M. Conversion of deletions during recombination in pneumococcal transformation. Genetics. 1989; 123: 455–464. PMID: 2599365
- 94. Adams A. Transformation and transduction of a large deletion mutation in *Bacillus subtilis*. Mol Gen Genet MGG.; 1972; 118: 311–322. doi: 10.1007/BF00333566 PMID: 4632051
- Cornejo OE, Rozen DE, May RM, Levin BR. Oscillations in continuous culture populations of *Strepto-coccus pneumoniae*: population dynamics and the evolution of clonal suicide. Proc R Soc B. 2009; 276: 999–1008. doi: <u>10.1098/rspb.2008.1415</u> PMID: <u>19129121</u>
- **96.** Engelmoer DJP, Donaldson I, Rozen DE. Conservative Sex and the Benefits of Transformation in *Streptococcus pneumoniae*. PLoS Pathog. 2013; 9: 1–7. doi: <u>10.1371/journal.ppat.1003758</u>

- 97. Li Y, Thompson CM, Trzciński K, Lipsitch M. Within-host selection is limited by an effective population of *Streptococcus pneumoniae* during nasopharyngeal colonization. Infect Immun. 2013; 81: 4534– 4543. doi: <u>10.1128/IAI.00527-13</u> PMID: <u>24082074</u>
- Chewapreecha C, Harris SR, Croucher NJ, Turner C, Marttinen P, Cheng L, et al. Dense genomic sampling identifies highways of pneumococcal recombination. Nat Genet. 2014; 46: 305–309. doi: <u>10.</u> 1038/ng.2895 PMID: 24509479
- **99.** Croucher NJ, Page AJ, Connor TR, Delaney AJ, Keane JA, Bentley SD, et al. Rapid phylogenetic analysis of large samples of recombinant bacterial whole genome sequences using Gubbins. Nucleic Acids Res. 2015; 43: e15. doi: 10.1093/nar/gku1196 PMID: 25414349
- 100. Rocha EP. Evolutionary patterns in prokaryotic genomes. Current Opinion in Microbiology. 2008. pp. 454–460. doi: <u>10.1016/j.mib.2008.09.007</u> PMID: <u>18838127</u>
- 101. Zhou Z, McCann A, Weill F-X, Blin C, Nair S, Wain J, et al. Transient Darwinian selection in Salmonella enterica serovar Paratyphi A during 450 years of global spread of enteric fever. Proc Natl Acad Sci. 2014; 111: 12199–12204. doi: 10.1073/pnas.1411012111 PMID: 25092320
- 102. Del Grosso M, Iannelli F, Messina C, Santagati M, Petrosillo N, Stefani S, et al. Macrolide efflux genes mef(A) and mef(E) are carried by different genetic elements in *Streptococcus pneumoniae*. J Clin Microbiol. 2002; 40: 774–778. PMID: <u>11880392</u>
- 103. Glaser P, Rusniok C, Buchrieser C, Chevalier F, Frangeul L, Msadek T, et al. Genome sequence of Streptococcus agalactiae, a pathogen causing invasive neonatal disease. Mol Microbiol. 2002; 45: 1499–1513. doi: <u>10.1046/j.1365-2958.2002.03126.x</u> PMID: <u>12354221</u>
- 104. Lopez-Sanchez MJ, Sauvage E, Da Cunha V, Clermont D, Ratsima Hariniaina E, Gonzalez-Zorn B, et al. The highly dynamic CRISPR1 system of *Streptococcus agalactiae* controls the diversity of its mobilome. Mol Microbiol. 2012; 85: 1057–1071. doi: <u>10.1111/j.1365-2958.2012.08172.x</u> PMID: 22834929
- 105. Deveau H, Barrangou R, Garneau JE, Labonté J, Fremaux C, Boyaval P, et al. Phage response to CRISPR-encoded resistance in Streptococcus thermophilus. J Bacteriol. 2008; 190: 1390–1400. doi: 10.1128/JB.01412-07 PMID: 18065545
- 106. Edgar R, Qimron U. The Escherichia coli CRISPR system protects from λ lysogenization, lysogens, and prophage induction. J Bacteriol. 2010; 192: 6291–6294. doi: <u>10.1128/JB.00644-10</u> PMID: 20889749
- 107. Goldberg GW, Jiang W, Bikard D, Marraffini LA. Conditional tolerance of temperate phages via transcription-dependent CRISPR-Cas targeting. Nature. 2014; 514: 633–637. doi: <u>10.1038/nature13637</u> PMID: 25174707
- 108. Holden MTG, Heather Z, Paillot R, Steward KF, Webb K, Ainslie F, et al. Genomic evidence for the evolution of *Streptococcus equi*: host restriction, increased virulence, and genetic exchange with human pathogens. PLoS Pathog.; 2009; 5: e1000346. doi: <u>10.1371/journal.ppat.1000346</u> PMID: <u>19325880</u>
- 109. Borezee E, Msadek T, Durant L, Berche P. Identification in *Listeria monocytogenes* of MecA, a homologue of the *Bacillus subtilis* competence regulatory protein. J Bacteriol. 2000; 182: 5931–4. doi: <u>10.</u> <u>1128/JB.182.20.5931-5934.2000</u> PMID: <u>11004200</u>
- Buchrieser C, Rusniok C, Kunst F, Cossart P, Glaser P. Comparison of the genome sequences of *Listeria monocytogenes* and *Listeria innocua*: clues for evolution and pathogenicity. FEMS Immunol Med Microbiol. 2003; 35: 207–213. PMID: 12648839
- 111. Levin PA, Margolis PS, Setlow P, Losick R, Sun D. Identification of *Bacillus subtilis* genes for septum placement and shape determination. J Bacteriol. 1992; 174: 6717–28. PMID: <u>1400224</u>
- 112. Attaiech L, Granadel C, Claverys JP, Martin B. RadC, a misleading name? J Bacteriol. 2008; 190: 5729–5732. doi: 10.1128/JB.00425-08 PMID: 18556794
- Jorth P, Whiteley M. An evolutionary link between natural transformation and CRISPR adaptive immunity. MBio. 2012; 3: 1–7. doi: 10.1128/mBio.00309-12
- 114. Liu F, Zhu Y, Yi Y, Lu N, Zhu B, Hu Y. Comparative genomic analysis of Acinetobacter baumannii clinical isolates reveals extensive genomic variation and diverse antibiotic resistance determinants. BMC Genomics. 2014; 15: 1163. doi: 10.1186/1471-2164-15-1163 PMID: 25534766
- 115. Redfield RJ, Findlay WA, Bossé J, Kroll JS, Cameron ADS, Nash JH. Evolution of competence and DNA uptake specificity in the Pasteurellaceae. BMC Evol Biol. 2006; 6: 82. doi: <u>10.1186/1471-2148-6-82</u> PMID: 17038178
- Martin B, Garcia P, Castanié M-P, Claverys J-P. The recA gene of Streptococcus pneumoniae is part of a competence-induced operon and controls lysogenic induction. Mol Microbiol.; 1995; 15: 367– 379. PMID: <u>7538190</u>

- Setlow JK, Boling ME, Allison DP, Beattie KL. Relationship between prophage induction and transformation in *Haemophilus influenzae*. J Bacteriol. 1973; 115: 153–161. PMID: 4541535
- 118. Yasbin RE, Wilson GA, Young FE. Transformation and transfection in lysogenic strains of *Bacillus subtilis*: evidence for selective induction of prophage in competent cells. J Bacteriol. 1975; 121: 296–304. PMID: 803952
- Beaber JW, Hochhut B, Waldor MK. SOS response promotes horizontal dissemination of antibiotic resistance genes. Nature. 2004; 427: 72–74. doi: <u>10.1038/nature02241</u> PMID: <u>14688795</u>
- 120. Auchtung JM, Lee CA, Monson RE, Lehman AP, Grossman AD. Regulation of a *Bacillus subtilis* mobile genetic element by intercellular signaling and the global DNA damage response. Proc Natl Acad Sci U S A. 2005; 102: 12554–12559. doi: 10.1073/pnas.0505835102 PMID: 16105942
- 121. Morrison DA, Guédon E, Renault P. Competence for natural genetic transformation in the Streptococcus bovis Group streptococci S. infantarius and S. macedonicus. J Bacteriol. 2013; 195: 2612–2620. doi: 10.1128/JB.00166-13 PMID: 23543718
- 122. Tong H, Zhu B, Chen W, Qi F, Shi W, Dong X. Establishing a genetic system for ecological studies of Streptococcus oligofermentans. FEMS Microbiol Lett.; 2006; 264: 213–219. doi: <u>10.1111/j.1574-6968.2006.00453.x</u> PMID: <u>17064375</u>
- 123. Zaccaria E, van Baarlen P, de Greeff A, Morrison DA, Smith H, Wells JM. Control of competence for DNA transformation in *Streptococcus suis* by genetically transferable pherotypes. PLoS ONE. 2014; 9: e99394. doi: 10.1371/journal.pone.0099394 PMID: 24968201
- 124. Beres SB, Sesso R, Pinto SWL, Hoe NP, Porcella SF, Deleo FR, et al. Genome sequence of a Lancefield group C Streptococcus zooepidemicus strain causing epidemic nephritis: new information about an old disease. PLoS ONE. 2008; 3: e3026. doi: 10.1371/journal.pone.0003026 PMID: 18716664
- 125. Nozawa T, Furukawa N, Aikawa C, Watanabe T, Haobam B, Kurokawa K, et al. CRISPR inhibition of prophage acquisition in *Streptococcus pyogenes*. PLoS ONE. 2011; 6: e19543. doi: <u>10.1371/journal.pone.0019543</u> PMID: <u>21573110</u>
- Bobay L-M, Rocha EPC, Touchon M. The adaptation of temperate bacteriophages to their host genomes. Mol Biol Evol. 2013; 30: 737–51. doi: <u>10.1093/molbev/mss279</u> PMID: <u>23243039</u>
- 127. Aziz RK, Ismail SA, Park H-W, Kotb M. Post-proteomic identification of a novel phage-encoded streptodornase, Sda1, in invasive M1T1 Streptococcus pyogenes. Mol Microbiol.; 2004; 54: 184–197. PMID: 15458415
- 128. Sakamoto JJ, Sasaki M, Tsuchido T. Purification and characterization of a *Bacillus subtilis* 168 nuclease, YokF, involved in chromosomal DNA degradation and cell death caused by thermal shock treatments. J Biol Chem. 2001; 276: 47046–51. doi: 10.1074/jbc.M106205200 PMID: 11584000
- 129. Gaasbeek EJ, Wagenaar JA, Guilhabert MR, Wösten MMSM, Van Putten JPM, Van Der Graaf-van Bloois L, et al. A DNase encoded by integrated element CJIE1 Inhibits natural transformation of *Campylobacter jejuni*. J Bacteriol. 2009; 191: 2296–2306. doi: <u>10.1128/JB.01430-08</u> PMID: <u>19151136</u>
- 130. Gaasbeek EJ, Wagenaar JA, Guilhabert MR, van Putten JPM, Parker CT, van der Wal FJ. Nucleases encoded by the integrated elements CJIE2 and CJIE4 inhibit natural transformation of *Campylobacter jejuni*. J Bacteriol.; 2010; 192: 936–941. doi: 10.1128/JB.00867-09 PMID: 20023031
- 131. Dalia AB, Seed KD, Calderwood SB, Camilli A. A globally distributed mobile genetic element inhibits natural transformation of *Vibrio cholerae*. Proc Natl Acad Sci. 2015; 112: 10485–10490. doi: <u>10.1073/</u> <u>pnas.1509097112</u> PMID: <u>26240317</u>
- 132. McMahon SA, Roberts GA, Johnson KA, Cooper LP, Liu H, White JH, et al. Extensive DNA mimicry by the ArdA anti-restriction protein and its role in the spread of antibiotic resistance. Nucleic Acids Res. 2009; 37: 4887–4897. doi: 10.1093/nar/gkp478 PMID: 19506028
- 133. Bondy-Denomy J, Pawluk A, Maxwell KL, Davidson AR. Bacteriophage genes that inactivate the CRISPR/Cas bacterial immune system. Nature. 2013; 493: 429–32. doi: <u>10.1038/nature11723</u> PMID: <u>23242138</u>
- 134. Croucher NJ, Mitchell AM, Gould KA, Inverarity D, Barquist L, Feltwell T, et al. Dominant role of nucleotide substitution in the diversification of serotype 3 pneumococci over decades and during a single infection. PLoS Genet. 2013; 9: e1003868. doi: 10.1371/journal.pgen.1003868 PMID: 24130509
- 135. Novick RP, Clowes RC, Cohen SN, Curtiss R, Falkow S. Uniform nomenclature for bacterial plasmids: a proposal. Bacteriol Rev. 1976; 40: 525.
- 136. Ramirez M, Severina E, Tomasz A. A high incidence of prophage carriage among natural isolates of Streptococcus pneumoniae. J Bacteriol. 1999; 181: 3618–3625. PMID: 10368133
- Labrie SJ, Samson JE, Moineau S. Bacteriophage resistance mechanisms. Nat Rev Microbiol. 2010; 8: 317–327. doi: 10.1038/nrmicro2315 PMID: 20348932

- 138. Croucher NJ, Finkelstein JA, Pelton SI, Mitchell PK, Lee GM, Parkhill J, et al. Population genomics of post-vaccine changes in pneumococcal epidemiology. Nat Genet. 2013; 45: 656–663. doi: <u>10.1038/</u> <u>ng.2625</u> PMID: <u>23644493</u>
- 139. Croucher NJ, Kagedan L, Thompson CM, Parkhill J, Bentley SD, Finkelstein JA, et al. Selective and Genetic Constraints on Pneumococcal Serotype Switching. PLoS Genet.; 2015; 11: e1005095. doi: 10.1371/journal.pgen.1005095 PMID: 25826208
- 140. Didelot X, Eyre DW, Cule M, Ip CL, Ansari MA, Griffiths D, et al. Microevolutionary analysis of *Clostrid-ium difficile* genomes to investigate transmission. Genome Biol. 2012; 13: R118. doi: <u>10.1186/gb-2012-13-12-r118 PMID: 23259504</u>
- 141. Hyatt D, Chen GL, Locascio PF, Land ML, Larimer FW, Hauser LJ. Prodigal: prokaryotic gene recognition and translation initiation site identification. BMC Bioinformatics. 2010; 11: 119. doi: <u>10.1186/</u> <u>1471-2105-11-119</u> PMID: <u>20211023</u>
- 142. Croucher NJ, Walker D, Romero P, Lennard N, Paterson GK, Bason NC, et al. Role of conjugative elements in the evolution of the multidrug-resistant pandemic clone *Streptococcus pneumoniae*-Spain23F ST81. J Bacteriol. 2009; 191: 1480–1489. doi: <u>10.1128/jb.01343-08</u> PMID: <u>19114491</u>
- 143. Camacho C, Coulouris G, Avagyan V, Ma N, Papadopoulos J, Bealer K, et al. BLAST+: architecture and applications. BMC Bioinformatics. 2009; 10: 421. doi: <u>10.1186/1471-2105-10-421</u> PMID: 20003500
- Li H, Durbin R. Fast and accurate short read alignment with Burrows–Wheeler transform. Bioinformatics. 2009; 25: 1754–1760. doi: 10.1093/bioinformatics/btp324 PMID: 19451168
- 145. Croucher NJ, Chewapreecha C, Hanage WP, Harris SR, McGee L, van der Linden M, et al. Evidence for Soft Selective Sweeps in the Evolution of Pneumococcal Multidrug Resistance and Vaccine Escape. Genome Biol Evol.; 2014; 6: 1589–1602. doi: <u>10.1093/gbe/evu120</u> PMID: <u>24916661</u>
- 146. Simpson JT, Durbin R. Efficient construction of an assembly string graph using the FM-index. Bioinformatics. 2010; 26: i367–73. doi: <u>10.1093/bioinformatics/btq217</u> PMID: <u>20529929</u>
- 147. Costa GG, Vidal RO, Carazzolle MF. Zorro [Internet]. 2011. http://www.lge.ibi.unicamp.br/zorro/
- 148. Carver T, Berriman M, Tivey A, Patel C, Bohme U, Barrell BG, et al. Artemis and ACT: viewing, annotating and comparing sequences stored in a relational database. Bioinformatics. 2008/10/11 ed. 2008; 24: 2672–2676. doi: 10.1093/bioinformatics/btn529 PMID: 18845581
- 149. Postingl H. SMALT [Internet]. 2012. http://www.sanger.ac.uk/resources/software/smalt/
- **150.** Kent WJ. BLAT—the BLAST-like alignment tool. Genome Res. 2002; 12: 656–664. doi: <u>10.1101/gr.</u> <u>229202</u> PMID: <u>11932250</u>
- 151. Boetzer M, Henkel C V, Jansen HJ, Butler D, Pirovano W. Scaffolding pre-assembled contigs using SSPACE. Bioinformatics. 2010/12/15 ed. 2011; 27: 578–579. doi: <u>10.1093/bioinformatics/btq683</u> PMID: 21149342
- 152. Fouts DE. Phage\_Finder: Automated identification and classification of prophage regions in complete bacterial genome sequences. Nucleic Acids Res. 2006; 34: 5839–5851. PMID: <u>17062630</u>
- 153. Lanie JA, Ng WL, Kazmierczak KM, Andrzejewski TM, Davidsen TM, Wayne KJ, et al. Genome sequence of Avery's virulent serotype 2 strain D39 of *Streptococcus pneumoniae* and comparison with that of unencapsulated laboratory strain R6. J Bacteriol. 2007; 189: 38–51. JB.01148-06 PMID: <u>17041037</u>